UNIVERSITY of York

This is a repository copy of *Mother's health after baby's birth: does delivery method matter?*:Does the delivery method matter?.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/141428/</u>

Version: Accepted Version

Article:

Tonei, Valentina (2018) Mother's health after baby's birth: does delivery method matter?:Does the delivery method matter? Journal of Health Economics. pp. 182-196. ISSN 0167-6296

https://doi.org/10.1016/j.jhealeco.2018.11.006

Reuse

This article is distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs (CC BY-NC-ND) licence. This licence only allows you to download this work and share it with others as long as you credit the authors, but you can't change the article in any way or use it commercially. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



eprints@whiterose.ac.uk https://eprints.whiterose.ac.uk/

Mother's mental health after childbirth: does the delivery method matter?

Valentina Tonei

DERS, University of York

August 5, 2018

Abstract

The dramatic increase in the utilization of caesarean section has raised concerns on its impact on public expenditure and health. While the financial costs associated with this surgical procedure are well recognized, less is known on the intangible health costs borne by mothers and their families. We contribute to the debate by investigating the effect of unplanned caesarean deliveries on mothers' mental health in the first nine months after the delivery. Differently from previous studies, we account for the unobserved heterogeneity due to the fact that mothers who give birth through an unplanned caesarean delivery may be different than mothers who give birth with a natural delivery. Identification is achieved exploiting exogenous variation in the position of the baby in the womb at the time of delivery while controlling for hospital unobserved factors. We find that mothers having an unplanned caesarean section are at higher risk of developing postnatal depression and this result is robust to alternative specifications.

Keywords: Caesarean Section, Instrumental Variables, Maternal Health, Millennium Cohort Study, Postnatal Depression.

JEL codes: I12, I18

Contacts: Valentina Tonei: valentina.tonei@york.ac.uk

I acknowledge financial support by the Economic and Social Research Council through my Ph.D. scholarship. I thank Gloria Moroni, Cheti Nicoletti, Luigi Siciliani and Emma Tominey for the ongoing support and useful suggestions; Kate Picket, Fiona Mensah and Holly Essex for providing Stata codes of some derived variables; participants at the Applied MicroEconometrics cluster seminars (York, 2014 and 2015), the 3rd WRDTC Economics Pathway PhD Conference (Sheffield, 2014), the iHEA World Congress (Dublin, 2014), CLS Cohort Studies Research Conference (London, 2015) and the ESPE Annual Conference (2017) for comments.

1 Introduction

Over the past few decades, a dramatic growth in the caesarean section (CS) rate has been recorded in many developed countries, regardless of the type of healthcare system (and relative incentives for physicians) and women's health needs (Bragg et al., 2010; Gibbons et al., 2010). In England, for example, the overall rate was about nine per cent during the 1980s, while nowadays more than one-fourth of women gives birth through caesarean delivery (Health and Social Care Information Centre, 2012; OECD, 2017). This implies that the incidence of this surgical procedure has almost tripled over the last 30 years. Similar patterns have been experienced by other OECD countries (Declercq et al., 2006; Macfarlane et al., 2015; OECD, 2015), raising questions regarding the economic implications of alternative delivery methods.

Concerns about the increase in caesarean section utilisation are justified by the higher economic and health costs associated with this procedure compared to a normal delivery (Koechlin et al., 2010). Indeed, while it is undeniable that caesarean deliveries have life-saving effects for mothers and children, especially for those who have concurrent health conditions (Gholitabar et al., 2011), it is also recognised that this procedure is very expensive, being the cost of a caesarean delivery between 66 and 88 percent higher than the cost of a natural delivery (Gruber and Owings, 1996; Petrou et al., 2002; Epstein and Nicholson, 2009). Besides the financial impact, the World Health Organization (WHO) has highlighted the association of this procedure with short- and long-term health risks for the mother (WHO, 2015).

This paper aims to contribute to the debate by analysing the causal impact of unplanned caesarean sections on first-time mothers' mental health after childbirth. While the negative effects for mothers' physical health in terms of longer postpartum recovery and prolonged pain are well-known (Lydon-Rochelle et al., 2001; Jensen and Wüst, 2015), less evidence is available on the effect on their psychological well-being. Mental health issues in general, and postnatal depression in particular, have been found to largely impact the mother's life, as well as being associated with a deterioration of her physical well-being and the relationship with her partner (Mutryn, 1993). Previous studies have also shown a strong link between maternal mental health and child development (Minkovitz et al., 2005; Propper et al., 2007; Kiernan and Mensah, 2009; Coneus and Spiess, 2012), rates of infections, hospital admissions and completion of recommended schedules of immunization for children (WHO, 2015), child health (Perry, 2008), long term child educational, labour market and criminal outcomes (Johnston et al., 2013).

For identification, we focus on unplanned deliveries. This is justified by two reasons. First, elective and unplanned caesarean deliveries may have different impacts on mother's mental health. Indeed, unplanned caesareans are unexpected, usually mentally and physically stressful, and associated with a loss of control and unmatched expectations (Soet et al., 2003, Lobel and DeLuca, 2007). On the contrary, planned caesareans are scheduled in advance, allowing for the possibility for women to adjust (at least partially) their expectations for this event. The second motivation concerns a limitation of the data employed in the analysis. We cannot distinguish among planned caesarean deliveries, those that have been scheduled because of the mothers' or babies' health needs and those requested by mothers for other reasons (the so-called *caesarean delivery on request*). Distinguishing between the two cases may be important because a different psychological impact of this procedure is expected depending on the reason why it has been implemented.¹

Most of the medical studies investigating the relationship between the delivery method and maternal mental health find that caesarean deliveries are expected to carry higher risks for mothers' mental health compared to natural deliveries. Indeed, women who have a caesarean delivery are more likely to suffer from physical pain after childbirth and have longer and more difficult postnatal recovery (Lydon-Rochelle et al., 2001, Cooklin et al., 2015), both conditions that also affect their psychological well-being. Additionally, caesarean deliveries may have a direct effect on mothers' mental health due to separation of mothers and their babies in the instants after the delivery (Mutryn, 1993). However, previous literature investigating this topic has not reached a unanimous consensus on whether having a caesarean delivery increases the risk of postnatal depression. This may depend on the limitations that characterise some of these studies, such as the small sample usually restricted to a particular geographic location or a population cohort, which does not allow to generalise the results to the entire population (Fisher et al., 1997; Koo et al., 2003). Failure to distinguish between elective (i.e. planned) and unplanned caesarean deliveries might also represent an issue, given that people tend to adjust better to traumatic events when they can predict or prepare for them (Clement, 2001). Additionally, the variability in the source of information on mothers' mental health (e.g. medical visits, self-completion questionnaires) and in the length of the postnatal period during which mothers develop depression (from a few weeks to one year after childbirth), can contribute to explain such heterogeneity (Robertson et al., 2004; Patel et al., 2005; Carter et al., 2006).

In general, these studies are characterised by the common assumption that the treatment (i.e. giving birth through an unplanned caesarean delivery) is randomly assigned. In other words, they implicitly assume that women who have an unplanned caesarean do not differ from those who give birth naturally except through observable characteristics for which we can control.² However, because of data limitations and the multiplicity of factors that can have an impact on both the delivery method and the mothers' mental health, it is very unlikely to be the case. As a result, the estimates reported in these studies may be (downward) biased.

 $^{^{1}}$ Results when considering both elective and emergency caesarean delivery are available on request from the author.

²This is equivalent to what Imbens and Wooldridge (2008) call *unconfoundedness assumption*, which assumes that adjusting for differences in observed pretreatment variables removes biases from comparisons between treated (mothers who give birth through unplanned caesarean deliveries) and control units (mothers who have a natural delivery).

This paper builds on the previous research by addressing some of these issues. It investigates the effect of caesarean deliveries on the risk of postnatal depression by employing a nationally-representative sample of mothers during their first pregnancy experience, obtained from the first sweep of the UK Millennium Cohort Study. We use a medium-term measure of maternal postnatal depression, which captures a period of sadness in the first nine months after childbirth. More importantly, this study represents the first attempt to identify the *causal* link between unplanned caesarean deliveries and mothers' mental health, by accounting for unobserved differences between mothers who give birth through different delivery methods (endogeneity). While other papers in the economic literature have dealt with this issue when analysing the causal effect of the delivery method on mothers' (Halla et al., 2016) and children's outcomes (Jensen and Wüst, 2015; Costa-Ramon et al., 2018), none of them have looked at the psychological consequences for mothers, yet the econometric methods they have employed are different from those used in this study.

We identify two main sources of endogeneity. One is due to the unobserved hospital characteristics that can both affect the choice of the delivery method as well as the risk of developing postnatal depression. For example, the level of resources available in a hospital in terms of staff and operating rooms may affect the level and standard of care. More specifically, in a hospital with a low nurse-to-patient ratio, women may receive less attention both during the labor and after the delivery. This may translate into (a) more compilations during labor, and therefore, into a higher risk of having an unplanned caesarean and (b) less psychological support after the delivery. At the same time, because the caesarean section is a surgical procedure, which requires it to be performed in operating rooms by surgeons, medical staff may decide to opt for this delivery method only in extreme cases and they may prefer to perform a natural (or instrumental) delivery whenever possible.

The second potential source of endogeneity is related to the fact that mothers who have an unplanned caesarean section might be systematically different from mothers who give birth naturally, in terms of their own health and of the health of their babies. While we can control for some characteristics (e.g. maternal age and baby's health at birth) which the literature has identified as driving the risk of having an unplanned caesarean delivery, there may be other factors we cannot observe due to data limitations, such as the mother's mental and physical health before and during the pregnancy. Also, we do not have detailed information on what occurred during the delivery and whether pain relief was used. However, epidural anesthesia has been found to increase the risk of unplanned caesarean delivery, and it is also associated with a reduction of pain during labour, which turns into a lower psychological impact from this event. As a result, failing to account for this factor can affect the estimation results.

In order to overcome these problems, we adopt an instrumental variable approach combined with hospital fixed effects, the latter being used to control for time-invariant characteristics at hospital level. As source

of exogenous variation we exploit, for the first time in the literature, the position of the baby in the womb at the time of delivery. It has been shown that, conditional on the mother's observable characteristics, the probability of having babies in an abnormal position (i.e. with shoulders or feet first) is random³ and mothers cannot affect it with their behavior (Impey et al., 2017).

Our results show that having an unplanned caesarean delivery increases the risk of postnatal depression. Without accounting for endogeneity, we find that a woman who gives birth through this procedure is 2.5 percentage points more likely to experience postnatal depression. The sign of this effect is confirmed by the IV estimates, even if in this case marginal effects are larger (15.1 percentage points). Results are robust to a number of specifications.

The rest of the paper is organized as follows. Section 2 summarises previous literature on this topic. Data are described in Section 3, while Section 4 presents our empirical strategies. Section 5 shows the main results and discusses the validity of the instruments employed in the analysis. Section 6 provides some sensitivity checks, while Section 7 concludes.

2 Related Literature

The worldwide increase in the utilisation of caesarean sections has attracted the attention of researchers to this phenomenon. Economists and health scientists have extensively investigated the medical and economic factors explaining this trend, together with its economic consequences for healthcare systems. Until now, however, only few papers exist that have focused on the impact of caesarean deliveries on mothers' health and even less have focused on their psychological well-being.

Most of the available evidence on the relationship between mode of delivery and maternal mental health after childbirth comes from the medical literature and shows mixed findings, making it difficult to draw useful conclusions from a policy perspective. The most robust evidence is provided by Patel et al. (2005), who employ a rich dataset, the Avon Longitudinal Study of Parent And Children, to analyse whether there are different incidence rates of postnatal depression among women who gave birth through different modes of delivery (elective caesarean, unplanned caesarean and natural deliveries). They measure postnatal depression eight weeks after childbirth and find no differences in the probability of experiencing postnatal depression among these groups of women. However, while this study relies on good-quality data, it fails to account for unobserved differences in these groups of women.

Fisher et al. (1997) employed a prospective study of 272 voluntary women and found that having a caesarean section does increase the risk of postnatal depression. However, because voluntary women are

³For a more discussion, see Section 4.

a selected group, this sample cannot be considered random and results cannot be generalised to the entire population.

Similar results were found by Koo et al. (2003), who used a retrospective comparative cohort study of 250 Malaysian women. They were interviewed at least six weeks after childbirth and found that, with respect to women having a non-emergency delivery, those who gave birth through caesarean section had about twice the risk of developing postnatal depression.

The meta-analysis by Carter et al. (2006) examined the association between caesarean section and postnatal depression, measured between 10 days and one year after delivery. The authors mentioned methodological weaknesses and the possibility that having a caesarean is a weak risk factor for postnatal depression as an explanation for the absence of clear evidence, suggesting that only high-quality studies may be able to identify such an effect.

Finally, in their review of the risk factors for postnatal depression, Robertson et al. (2004) classified caesarean section as a weak determinant. They drew this conclusion because of the mixed evidence from previous research with Warner et al. (1996), Forman et al. (2000) and Johnston et al. (2013), failing to find any relevant link. However, Boyce and Todd (1992) and Hannah et al. (2002) found a highly significant association between this delivery method and postnatal depression after three months and six weeks, respectively.

Economic studies analysing the health consequences of caesarean section have instead focused mostly on the baby's health (e.g. Jensen and Wüst, 2015; Costa-Ramon et al., 2018) and the mother's fertility and labour decisions (Halla et al., 2016; Card et al., 2018).

3 Data

The data comes from the UK Millennium Cohort Study (MCS), a multidisciplinary longitudinal data set on a cohort of children born between September 2000 and January 2002. This data set covers several topics, such as parenting, childcare, child behaviour and cognitive development, child's and parental health, pregnancy and delivery, parents' employment and education, as well as income and poverty. Information is derived from the parents' interviews in six sweeps, when children are 9-11 months, 3, 5, 7, 11 and 14 years old. For the purpose of this study, we use only the first sweep, which contains detailed information on circumstances of pregnancy and birth, as well as socio-economic background and the health conditions of the family where children were born.⁴

⁴For full details of the study, see Plewis et al. (2007).

3.1 Sample selection

The initial sample is characterised by 18,818 children, born from 18,552 women. We focus on natural mothers (99.9% of the sample) and we exclude women who had a multiple delivery (256) because they are more likely to have had health complications after childbirth and their babies are systematically different in terms of (lower) birth weight, gestational age at birth and other birth characteristics in comparison to singletons. In addition, since we use information on the place of the delivery and particularly hospital identifiers, we exclude deliveries that occurred at home or in unknown hospitals (384). We also drop observations with missing or incomplete information on the delivery method, mother's health, pregnancy, demographic characteristics and baby's health (leaving us with 84.51% of the original sample). We drop observations if the baby was delivered through a planned caesarean delivery or pre-term⁵ (1,458) and the woman had previous pregnancies (7,472). Our final sample includes 5,896 women.

3.2 Variables

The outcome variable is a binary indicator that takes a value equal to one if the mother reports to have experienced a period of sadness lasting two weeks or more after childbirth. Previous literature focusing on the effect of caesarean delivery on mother's mental health shows a high degree of heterogeneity in the definition of postnatal depression⁶. In particular, while there is a general agreement in the medical community on the symptoms that identify postnatal depression (e.g. low mood, loss of enjoyment and pleasure, anxiety) and the length of the period after delivery⁷, the time of onset that defines postnatal depression is less clear.⁸ In this paper, we follow the definition of postnatal depression suggested by Mcintosh (1993), by considering postnatal depression as the experience of a depressed mood for a period of at least two weeks at some stage during the first nine months after delivery. Compared to other measures, this can be considered a medium- to long-term indicator of maternal postnatal depression. However, MCS does not provide details on the severity of this condition; therefore, mothers reporting symptoms of postpartum depression may be affected by this condition differently.

A potential limit of this measure is that it is built using self-reported information. The general concern about this type of health outcome is that it can measure individuals' health with error, being affected by unintentional (e.g. recall bias) and intentional bias (stigma associated with mental disorders may lead mothers to under-report mental illnesses). Nonetheless, more 'objective' measures, such as postnatal

 $^{^{5}}$ We define pre-term births if they occurred before the 37th week of gestation.

 $^{^{6}}$ For example, Patel et al. (2005) measured postnatal depression about two months after the delivery. In their metaanalysis, Carter et al. (2006) reviewed studies that measured postnatal depression between 10 days and one year after the delivery.

 $^{^{7}}$ To be classified as postnatal depression the condition should last at least two weeks (Gavin et al., 2005).

⁸Some experts in the field define postpartum depression as a condition characterised by low mood and anxiety that occur anytime within the first year postpartum irrespective of the time of onset (Stuart-Parrigon and Stuart, 2014).

depression diagnosed by doctors, are not necessarily more appropriate. A report from the Royal College of Obstetricians and Gynaecologists shows that about half of the mothers who experienced mental health problems were not referred to services or offered any further information about where to go for support. Additionally, the probability of being diagnosed with depression depends on the frequency of the contacts with her GP. If the woman tends not to attend GP visits, there exists a risk of underestimating the incidence of this condition.

In the MCS, the mode of delivery is initially coded in ten categories (including missing and refused responses). We reclassify this measure into three mutually exclusive groups: natural, elective caesarean and unplanned caesarean delivery. Natural deliveries are defined as those that can be classified as medical procedures according to the Healthcare Resource Groups (HRG) system (therefore, this category also includes instrumented deliveries). Caesarean sections are distinguished as elective and unplanned, with the latter usually associated with unexpected complications at the time of delivery. Following Essex et al. (2013), for those women who reported more than one mode of delivery, we combine responses by coding the most invasive as the primary delivery method.⁹ Since the analysis focuses on the effect of unplanned caesarean deliveries compared to natural births, we drop from the sample elective caesareans and we define a binary variable, CS_m , taking a value equal to one if the woman had an unplanned caesarean delivery and zero in the case of a natural delivery.

There exists a variety of factors that may be associated with both the probability of experiencing postnatal depression and giving birth through unplanned caesarean delivery. We classify them in the following categories: (i) socio-economic factors, (ii) pregnancy-related attitude, (iii) maternal health, and (iv) baby's health at birth.

Among the socio-economic factors, mother's age is one of the most important. Teenagers are, on average, more likely to experience postnatal depression (Brown, 1996; Deal and Holt, 1998; Robertson et al., 2004). A set of dummies defining the mother's ethnicity are included to account for the heterogeneous composition of the UK population. In particular, we distinguish between (a) white mothers (either British or non-British), (b) black, (c) Pakistani, Bangladeshi and Indian, (d) other ethnicities (in particular Chinese and other Asian groups).¹⁰ Some ethnicity groups have been found to be less likely to be affected by anxiety and/or depression, probably because of their propensity to under-report health problems (Fiscella et al., 2002; Harris et al., 2005). Household income and the mothers' level of education have been identified by the economic literature as strong predictors of psychological wellbeing. They are also positively associated with unplanned caesarean sections, even after controlling for standard covariates

 $^{^{9}}$ We consider unplanned caesarean sections as the most invasive procedure, while natural births as the less invasive.

¹⁰This classification follows the Office for National Statistics (ONS) recommendation on how to classify ethnicity in the UK. The residual category includes Chinese and mixed ethnicity women. I cannot further split the residual category because of the small numbers of women belonging to these ethnicity groups in the sample.

(Gresenz et al., 2001; Segre et al., 2007). We measure the mother's highest level of education with a set of dummies that indicate her highest national vocational qualification. Marital status is included as a proxy for perceived social support. A married woman is expected to have psychological support from her husband in taking care of the child, which, in turn, reduces her likelihood of becoming depressed (Stewart et al., 2003).

We include an indicator measuring if the pregnancy was planned, as this may be correlated with the psychological impact of the delivery method. Physical and mental health before pregnancy are strong predictors of postnatal depression. Unfortunately, being that the MCS is a child-focused dataset, it does not include any direct information on maternal health before the child was born. To proxy her physical health before pregnancy, we use information about the mother's body mass index (including both a linear and a quadratic term to allow for non-linearities in the effect). Admission to hospitals and whether she had a paid job during pregnancy are included to control for the mother's health during childbearing. Furthermore, we include dummies to measure whether she has ever suffered from diabetes, gestational hypertension or kidney diseases.¹¹

According to the literature (e.g. McLennan et al., 2001), mothers of unhealthy children are negatively affected by their babies' health and, as a consequence, they are more likely to report a depression status. We account for this association by including binary indicators for underweight and overweight babies at birth and gestational age (in weeks) as measures of the baby's health.

Finally, our empirical approach exploits the exogenous variation in the position of the baby at birth. The instrument we derive is a binary variable, *Pos*, which takes the value of one if the baby was in a breech position or in another abnormal position requiring a surgical intervention.

Table 1 provides some descriptive statistics of the variables included in the analysis. 32.5 per cent of the women in the sample have experienced a period of sadness lasting at least two weeks after delivery. This percentage is above the documented prevalence rate for postnatal depression in the U.K., estimated to be around 15 per cent. However, this difference could be ascribed to variability in the severity of the symptoms¹² and in the length of the postnatal period considered. Additionally, previous literature has shown that higher rates of incidence are observed when employing a non-clinical definition of depression (e.g. self-reported measures as the one employed in this study), compared to the case in which standard instruments for the detection of depression are used¹³.

 $^{^{11}}$ For this last set of conditions, we only know whether the mother has ever suffered from these conditions. In other words, we do not have information on the time of the onset (if ever).

 $^{^{12}}$ Stuart-Parrigon and Stuart (2014) argued that depressive symptoms after the baby's birth affect more than 25% of new mothers, while the rates of major depressive disorders during the perinatal period range from 10-15%.

¹³A recent study used self-reported information and established that the incidence of postnatal depression in the UK is likely to be higher than previously estimated, with around 3 out of 10 mothers experiencing this condition, (NICE 2011).

The proportion of mothers who had an unplanned CS is 19.05 per cent, while 7.06 per cent had a baby in a breech position.¹⁴

On average, mothers are 25 years old and half of them are married. In terms of their health during pregnancy, 19 per cent were hospitalized and 9 per cent suffered from pre-eclampsia. The average gestational age is 40 weeks and the baby's birth weight is 3.4 kilos.

4 Empirical Strategies

We study the effect of the mode of delivery on the mother's mental health after childbirth. In particular, the empirical model explaining the risk of postnatal depression is specified following the standard health capital models as theorized by Grossman (1972) and firstly estimated by Rosenzweig and Schultz (1983). We adapt such a framework by defining a maternal mental health production function that includes medical, as well as socio-economic factors associated with the risk of postnatal depression.

$$PD_m = f(CS_m, \mathbf{SES_m}, \mathbf{X_m}, \mathbf{H_m^H}, \mathbf{H_m^C})$$
(1)

The outcome variable is PD_m , a binary variable denoting whether mother m suffered from postnatal depression in the first nine months after childbirth. CS_m represents the causal variable of interest and indicates whether the mother gave birth through an unplanned caesarean delivery (as compared to a natural delivery). SES_m is a vector including all socio-economic variables that may be related to the mother's mental health, such as age, marital status, income, ethnicity and level of education. X_m includes information on whether the pregnancy was planned. Measures of the mother's physical health, H_m^M , which refers to different points of time, are added to control for pre-existing health conditions. Finally, H_m^B

The health production function (Equation 1) is assumed to be additive separable and linear in its inputs:

$$PD_m = \beta_0 + \beta_1 CS_m + \mathbf{SES}_m \beta_2 + \mathbf{X}_m \beta_3 + \mathbf{H}_m^M \beta_4 + \mathbf{H}_m^C \beta_5 + \epsilon_m$$
(2)

where ϵ_m denotes the mother-specific idiosyncratic error.

 $^{^{14}}$ This proportion is larger than the national statistics because the sample we use includes first time mothers only. As mentioned above, the frequency of breech decreases with increasing parity because after the first pregnancy the womb tend to be larger, allowing the baby to move. When using the full sample, the proportion of breech babies decreases to 5 per cent.

¹⁵Gestational age at birth and birth weight may be endogenous as they are potentially affected by the mode of delivery. For this reason, as a sensitivity, we estimate the model excluding these variables from the vector of covariates. Results are reported in Section 6.

As previously noted, because the delivery method is not randomly assigned, a simple regression of CS_m on the indicator of postnatal depression is likely to lead to inconsistent estimates of the effect of the mode of delivery on the mother's mental health. This is also supported by the evidence provided in Table 2 where we compare mothers' characteristics by treatment (i.e. mode of delivery). The difference in socio-economic and medical characteristics of mothers who gave birth naturally compared to those who had an unplanned CS suggests that the two groups of mothers differ substantially and are likely to differ in unobservable characteristics as well.

In order to overcome these issues and obtain consistent estimates we combine two econometric approaches: hospital fixed effect models and instrumental variables.

4.1 Hospital fixed effects model

Hospital characteristics, such as internal organisation, resources availability (e.g. medical staff and operating rooms) and quality of care might affect both the probability of giving birth through an unplanned CS and the risk of developing postnatal depression. We control for these (time invariant) unobservable factors at hospital level by including hospital fixed effects into the model.

Equation 2 is rewritten to include hospital fixed effects:

$$PD_{mj} = \beta_0 + \beta_1 C S_{mj} + \mathbf{SES}_{mj} \beta_2 + \mathbf{X}_{mj} \beta_3 + \mathbf{H}_{mj}^{\mathbf{M}} \beta_4 + \mathbf{H}_{mj}^{\mathbf{B}} \beta_5 + \mu_j + \xi_{mj}$$
(3)

where PD_{mj} indicates whether mother *m* who gave birth in hospital *j* has suffered from postnatal depression in the first nine months after the baby's birth. This equation differs from Equation 2 because the error term is split into two components, an idiosyncratic mother-level component, ξ_{mj} , and an unobserved heterogeneity component at the hospital level, μ_j . By estimating this model, we get rid of the time-invariant unobserved heterogeneity at hospital level.

4.2 Instrumental variable approach

While the inclusion of hospital fixed effects accounts for differences in (time invariant) hospital characteristics, there may still be mother-specific unobservable characteristics correlated with the mode of delivery and their mental health (e.g. mother's health status during pregnancy and delivery experience), which would bias the results. We account for this issue by adopting an instrumental variable approach that exploits, as source of variation, the position of the baby in the womb before the delivery.

We define a binary variable, POS_m , equal to one if the baby presents feet or shoulders first, head at the back or other abnormal positions at birth, a situation called *breech position*. Full breech position at term means that the baby has not turned head down in the womb by week 37 of the pregnancy. If the baby is in a breech position at the end of week 36, the mother can be offered an external cephalic version (ECV). This is when a healthcare professional tries to turn the baby into a head-down position by applying pressure on the abdomen. The choice of performing this procedure depends on some health conditions of the mother (that we control for in the model).

If the baby remains in a breech position, the mother is given the option of a caesarean section or a natural delivery. Thus, the caesarean section can be either planned or unplanned, the latter case occurring in case of an emergency before she goes into labour (or before the planned CS). According to NICE statistics, 40% of women planning a natural breech birth end up needing an emergency caesarean section.

While breech presentation is very common in early pregnancy, by 36-37 weeks of pregnancy most babies turn into the head-first position, with 3-4% of babies being in the breech position (Royal College of Obstetricians and Gynaecologists, 2006) at the time of birth.

We define the first stage regression as:

$$CS_m = \alpha_0 + \alpha_1 Pos_m + \mathbf{SES_m}\alpha_2 + \mathbf{X_m}\alpha_3 + \mathbf{H_m^M}\alpha_4 + \mathbf{H_m^B}\alpha_5 + \epsilon_m$$
(4)

The instrumental variable approach requires two conditions to be met.

The first consists of the instrument being uncorrelated with ϵ_m , the error term (exclusion restriction or validity condition). In other words, having a baby in a breech position at the time of delivery must be uncorrelated with unobserved characteristics of the mother (and her pregnancy) that can also have an effect on her mental health. Tharin et al. (2011) argue that breech babies can be considered as a good random subgroup of all babies since there is no clear evidence of maternal or baby's characteristics that can predict the probability of a breech position. This is also supported by Jensen and Wüst (2015), who show that breech and non-breech mothers are similar in a range of observable characteristics, such as level of education and pregnancy conditions unrelated to breech (e.g. pre-eclampsia and diabetes). As well, the Royal College of Obstetricians and Gynaecologists states that, while persistent breech presentation may be associated with biological factors (such as amniotic fluid volume, the placental localisation and the uterus), it may be due to chance as well (Royal College of Obstetricians and Gynaecologists, 2006). This result is also generally confirmed by the medical literature, although there is some evidence of a weak association between the position of the baby and some predictors of postnatal depression, such as parity¹⁶, age and birth weight (Rayl et al., 1996; Fruscalzo et al., 2014).

¹⁶Parity is likely to be a predictor of breech position because after the first pregnancy, the womb tends to be larger, allowing the baby to move, and thus, reducing the risk of a breech position. In this study, we control for parity by focusing on first-time mothers. Controlling for the number of previous pregnancies in the model rather than restricting the analysis on first-time mothers would not solve this problem, as we would not be able to account for the heterogeneity in previous

While the validity condition cannot be tested, we provide some evidence in support of using breech position as an excluded instrument.

We first regress the baby's position on a full set of the mothers' and baby's covariates to identify factors predicting the position of the baby. The idea is that if the we find no association between these factors and the baby's position, we have an argument in favour of the validity (exogeneity) of the instrument. Results show that none of the factors considered predicts the probability of having a baby in a breech position except the mother's BMI. This means that, once we control for BMI, the baby's position in the womb is exogenous. Supportive evidence of the baby's position is confirmed by the F-statistic for the joint significance of all the coefficients being very small (Table 3).

We also perform a balance test comparing the covariates of women whose babies were in a breech position with those of women whose babies were not in such a position. Table 4 reports the means for each variable by breech position. The majority of the variables considered do not differ in the two groups, in particular the baby's health at birth (proxied by birth weight and gestational age) and the mother's previous health conditions (e.g. diabetes, hypertension, eclampsia, etc.) are not associated with the baby's position in the womb. Consistently with the previous literature (e.g. Fruscalzo et al., 2014), we find that age is associated with the breech position. Similarly, income and education seem to increase the probability of having a baby in a breech position. However, these variables are likely to capture the age effect; indeed, we find that they are all strongly correlated with maternal age (for example, the correlation between age and income is 0.57, and its associated p-value is below 0.001). Being Pakistani, Bangladeshi or Indian is associated with a lower chance of having a baby in a breech position. This again can be explained by the fact that mothers belonging to this ethnic group are on average younger when they have the first child. Finally, we find some slight differences for BMI and being employed during the pregnancy. However, these differences are very small in magnitude. Therefore, all together this seems to suggest that breech position is random, once we control for the mother's age at birth.

So far, we have shown that there is no substantial evidence of selection on observables. However, what we really worry about is that women with breech and non-breech babies are different in terms of unobservable characteristics. We show that this is not the case by using the residuals obtained from a model that predicts the probability of having a baby in a breech position on the covariates included in the main model

pregnancy experiences (e.g. mode of delivery, number of previous pregnancies, etc.). The delivery method in previous pregnancies, for example, is an important factor that we cannot control for, because of data limitations. However, we know that the mode of delivery in previous pregnancies affects the delivery mode in subsequent deliveries (once a woman has had a caesarean section, the probability of having another caesarean is higher). Additionally, the impact of an unplanned delivery on the mothers' mental health can be different for women who had previous pregnancies because of the additional information they may already have on the birth experience. In Section 6, we show that results do not differ when we focus on all mothers, regardless of their previous pregnancy experiences. However, to guarantee the validity of the instrument, in the paper we focus on the case of first-time mothers.

to analyse whether the variation captured by the residuals is associated with unobservable characteristics of the mother. As a proxy for such unobservables we use information on the mother's smoking behaviour, which the literature has found to be strongly correlated with measures of human and social capital (Scheffler et al., 2010). We measure the mother's smoking behavior before the pregnancy, both in terms of the number of cigarettes (a measure for intensity), as well as whether the mother was a smoker. Table 5 shows that there is no statistically significant association between the residuals and our measures of unobserved skills and health, suggesting that the variation of our instrument is exogenous.

Another potential concern for the validity of the identification strategy employed in this study would be if mothers can affect the position of her baby (e.g. if mothers who know their babies are in a breech position try to change the position of the baby) and such behaviour is correlated with unobservables, which also affect the mothers' psychological well-being. We have already shown that there is no empirical evidence of a correlation between breech position and the mother's (observed and unobserved) characteristics. Additionally, the Royal College of Obstetricians and Gynaecologists has established that there is no scientific evidence to support the effectiveness of exercises performed by mothers to change the position of their baby in the womb (Impey et al., 2017). This means that, even if mothers try to change their baby's position by lying or sitting in a particular position, this behaviour does not actually change the probability of giving birth to a baby in a breech position. As a result, this is not a threat to the identification strategy.

The second condition to satisfy is the relevance of the instrument (i.e. baby's position in the womb), which must be a strong predictor of the mode of delivery (the endogenous variable). The National Institute of Clinical Excellence (NICE) encourages resorting to the caesarean delivery if a breech position occurs at the end of the gestational period in other to reduce the risk of perinatal mortality and neonatal morbidity (Gholitabar et al., 2011). Similar guidelines have been issued in other countries, especially after the publication of the results from the Term Breech Trial, the largest randomized control trial evaluating adequate modes of delivery for breech babies (Hannah et al., 2002; Whyte et al., 2004; Goffinet et al., 2006). As a result, a large proportion of breech babies are born through a caesarean delivery in the United Kingdom every year (Bragg et al., 2010), and similar rates are observed in other countries, e.g. U.S.A. (Lee et al., 2008), Sweden (Alexandersson et al., 2005), Denmark (Jensen and Wüst, 2015) and the Netherlands (Rietberg et al., 2005).¹⁷ The strong compliance with these guidelines that have been observed in many countries (including the United Kingdom) guarantees that the relevance condition is satisfied.

 $^{^{17}}$ However there are still countries, such as Belgium and France, where it is still common practice to perform natural deliveries even in case of breech babies. Goffinet et al. (2006) shows that in those countries, this is still a valid option, as the risk of complications for babies is very low.

4.3 Estimation methods

We implement an instrumental variable approach by adopting a two-stage least squares estimation using Pos_m as an instrument and including hospital fixed effects in the model.

The main advantages of using linear IV methods concerns the possibility of testing the relevance of the instrument and to interpret the coefficients in terms of marginal effects. However this specification ignores the binary nature of the health outcome and the endogenous variable. As a robustness check, we estimate bivariate probit models (Heckman, 1978; Nichols, 2011) and we adopt a two-stage residual inclusion approach (Terza et al., 2008). Results are shown and discussed in Section 6.

5 Results

Equation 2 is initially estimated with Ordinary Least Squares (OLS), treating the mode of delivery (CS) as exogenous. This specification can be viewed as a descriptive regression which sheds light on whether the effect of unplanned caesarean delivery persists after controlling for other observed factors. Also, it provides a benchmark against which to compare the results from fixed effects and IV models.¹⁸ Table 6 presents OLS estimates obtained by adding gradually different sets of covariates in the regression equation. The first column shows results when the mode of delivery and the socio-economic variables are included. In Column 2, we also include information on maternal health conditions and pregnancy. Finally, we add measures of the baby's health at birth to account for the negative impact of the poor baby's health on maternal mental health (Column 3).

In all the specifications, we find a positive and significant association between unplanned caesarean delivery and postnatal depression. When including measures of maternal health and pregnancy experience, the magnitude of the coefficient slightly decreases (moving from 0.029 to 0.025) and becomes less statistically significant, suggesting that part of the association is captured by the poorer levels of health of the mothers who gave birth through this procedure. The inclusion of measures of the baby's health at birth does not reduce the magnitude. Overall, when controlling for the full set of regressors, we find that having an unplanned caesarean delivery is associated with an increase in the likelihood of postnatal depression by 2.5 percentage points.

As discussed above, OLS estimates may be biased if there are omitted variables at the mother or hospital level. We first control for time-invariant hospital factors by including the hospital fixed effects in the model. Results, reported in Column 4 of Table 6, are not statistically different from those obtained in

¹⁸From a theoretical point of view, the linear probability model is not the appropriate specification, given the binary nature of the outcome variable. To account for this, we also estimate Equation 2 by adopting a probit model specification. Results, available from the author, confirm the OLS estimates.

the OLS specifications. This has at least two interpretations. First, it may be that unobservable hospital characteristics affecting both the delivery method and the risk of developing postnatal depression are not an issue in this context. Another explanation is that what matters to explain the risk of developing postnatal depression is the relationship of the mother with the nurse that follows her during and after the pregnancy, rather than factors at the hospital level, such as quality of care and resources available. Under the second scenario, hospital FE would not solve the endogeneity issue, being that this method is only able to account for hospital characteristics that do not vary over time and across mothers. However, the IV strategy can represent a solution, accounting for any source of endogeneity, provided that the instruments employed are valid.

5.1 First stage results

The first stage of linear IV models (Columns 1 and 3 of Table 7) estimated using Pos_m as an exclusion restriction, shows that the partial correlation between the baby's position in the womb and an unplanned CS is equal to 0.310 (0.314 when hospital fixed effects are included), and it is strongly statistically significant. This is in line with what obtained when comparing the proportion of breech babies born through an unplanned caesarean section with those who are born naturally, which are 0.197 and 0.048, respectively. Further evidence of the instrument relevance is provided in the bottom panel of Table 7. The F-statistic testing for the significance of the excluding restriction is equal to 163 (167.3 when hospital fixed effects are included)¹⁹ and the test on the weak identification of the IV model (robust Kleibergen-Paap Wald rk F statistic) suggest that the position of the baby in the womb is a strong predictor of the mode of delivery.

Interestingly, the estimated effects of the covariates are in line with other studies (Halla et al., 2016). Ethnicity is the most important socio-economic predictor, with black and Pakistani, Bangladeshi and Indian women being at higher risk of having an unplanned caesarean delivery. There is also a negative educational gradient, with more educated mothers being less likely to have an unplanned CS. Most of the measures of the mother's health before and during the pregnancy also have predictive power. For example, hospitalisation during pregnancy or having suffered from diabetes increases the risk of an unplanned caesarean delivery. We also find that underweight and overweight babies are more likely to be born through an unplanned CS. This is explained by the fact that an underweight baby may suffer from health conditions that affect their development and require a medical intervention. Instead, overweight babies are usually associated with post-term deliveries, which are more likely to be affected

 $^{^{19}}$ Stock and Yogo (2005) suggest 10 as a rule of thumb. Under such value, the relevance of the instrument(s) is not guaranteed.

by complications. Similar findings are obtained when we include hospital fixed effects to the model (see Column 3 in Table 7).

5.2 Second stage results

Column 2 and 4 of Table 7 report the second-stage results. We find that an unplanned caesarean delivery increases the risk of developing postnatal depression by 15.1 (18.8) percentage points when we include (exclude) hospital fixed effects. The effect is not negligible, given the underlying risk of postnatal depression of 32.5 per cent.

When focusing on the coefficients associated with the covariates included in the model, we find that most of them behave as expected. For example, married women have 5 percentage points lower of a chance of being affected by postnatal depression than unmarried women. Income also shows a strong negative association with the probability of experiencing postnatal depression (negative coefficient equal to 0.003). This result is consistent with the *family stress model*, as defined by Conger et al. (2000), which states how economic hardship and pressure negatively impact parents' mental health.

Among the health and pregnancy variables, we find that poor physical health, measured by hospitalization during pregnancy, strongly predicts postnatal depression and, coherently with previous studies that show a strong association between physical and mental health (Canadian Mental Health Association, 2008). Along the same lines, we find that working during pregnancy is negatively associated with the probability of developing depression after childbirth (negative coefficient equal to 0.044). On the contrary, diabetes and hypertension are not significantly associated with postnatal depression.

Results also show that having planned the pregnancy in advance is associated with a decrease in the probability of postnatal depression by 6.1 percentage points. Finally, poor baby's health (proxied by birth weight and gestational age) does not appear to have an effect on the probability of developing postnatal depression.²⁰

5.3 Interpretations of the results and analysis of the effect for low-risk pregnancies

Overall, we find that failing to account for selection into the delivery method leads to an underestimation of the impact of unplanned caesareans on maternal mental health. But, is this really the case? An explanation for our results may be that the linear IV model produces upward biased results because the instrument employed in the analysis does not satisfy the exclusion validity assumption. However, as we

 $^{^{20}}$ It is worth remembering that we have dropped from the sample pre-term births, meaning that the most severe cases (babies born before the 37th week) are not included.

discussed in Section 4, this is unlikely to be the case, as the tests we performed supports the validity of the instruments, so we rule out this hypothesis. Another explanation is that, if treatment effects are heterogeneous across the population of first-time mothers, what we obtain when exploiting the exogenous variation in the position of the baby in the womb is a local average treatment effect (LATE), i.e. the average treatment effect for a defined *subgroup* of women who had an unplanned caesarean delivery as a consequence of the position in the womb of their baby at birth (the so-called compliers, see Angrist and Pischke, 2008). That said, the effect would still be interesting and policy relevant, since breech babies account for around 4 percent of all births, which translates into about 30,000 babies born every year just in the UK alone. Similar proportions are observed in other countries (e.g. Denmark, see Jensen and Wüst, 2015).

If the effect we identify applies to mothers with breech babies only, a group of high risk deliveries, we may further wonder about the impact of an unplanned CS on mothers who had a low-risk delivery. We have tried to answer this question by following two approaches.

The first consists in identifying low-risk deliveries by excluding any cases whether the mother suffered (before or during the pregnancy) from medical conditions, which may affect her health or the health of her baby during the pregnancy and increase the risk of complications (such as diabetes, pre-eclampsia, kidney diseases) and pre-term deliveries. We also exclude mothers whose babies were in a breech position at the time of birth. By focusing on non-breech babies, we cannot implement the instrumental variable approach. Therefore, we estimate a hospital fixed effect model, similar to the one described in Equation 3. Results, reported in Table 8, do not differ from those obtained using the full sample (Column 4 in Table 6), suggesting that the effect of unplanned caesareans on mother's mental health is the same for high- and low- risk deliveries.

An alternative approach to identify the effect of unplanned CS on the mother's mental health in the case of low-risk CS would be to use the time of the day when the delivery occurred as an excluded instrument. Costa-Ramon et al. (2018) argued that "while nature distributes births and associated problems uniformly, time-dependent variables related to physicians' demand for leisure are significant predictors of unplanned c-sections". The idea is that, if the proportion of unplanned CS varies during the day (for example, if the rate of caesarean deliveries is higher during the early hours of the night) that would suggest that these procedures are performed following physicians preferences, and therefore cannot be classified as real emergencies. We test whether this is the case in our paper by estimating the first stage of a linear IV model using the time of the delivery (measured in hours) as an instrument. A strong correlation between the probability of having an unplanned CS and the time of the day would imply that we can identify low-risk (unnecessary) unplanned CS. Instead, we find a very weak association between the time of the delivery and the probability of having an unplanned caesarean, both when time is defined as a continuous measure and as a set of dummies (Table 9). These results indicate that these unplanned caesareans were medically-indicated, i.e. there was no physician's discretion on the choice of the mode of delivery and they had to be performed independently of the time of day.

Finally, we are interested in understanding whether the effect we observe is specific for first-time mothers or it can be also generalised to mothers who had previous pregnancies. We investigate this issue comparing our main results (reported in Table 7) with those obtained using a sample that includes first-time mothers as well as multiple-time mothers. We find no statistically significant differences in the impact of unplanned caesarean deliveries for the two groups (Table 10). Therefore, we conclude that having an unplanned caesarean delivery carries a lot of stress, regardless of the previous delivery experience.

6 Sensitivity checks and heterogeneity analysis

6.1 Model specification

Linear IV models (estimating by 2SLS) may not be the most appropriate model specification because they assume that the outcome and the endogenous regressor are continuous, while both CS_m and PD_m are binary indicators. We account for this issue by estimating the model using alternative approaches such as (a) a bivariate probit model and (b) two-stage residual inclusion (2SRI). Both these approaches account for the fact that the treatment (CS) and the outcome variables are binary.

The bivariate probit model assumes that the outcome and treatment are each determined by latent linear index models with jointly normal error terms. The two-stage residual inclusion (2SRI) estimation is a semi-parametric approach that uses the residuals from the first stage to control for endogeneity in the second stage (Terza et al., 2008; Basu et al., 2018). The estimates derived from a bivariate probit are interpreted as average treatment effects (ATE), as bivariate probit and other models of this sort can be used to estimate unconditional average causal effects and/or effects on the treated.

It is an open question whether we should rely on the linear IV model, the bivariate probit model, the 2SRI or other approaches. The existing literature offers conflicting advice. For example, Angrist and Pischke (2008) argue that "the hard part of the empirical problem is finding a good instrument and that the difficulties with endogenous variables in nonlinear limited dependent variables models are usually more apparent than real". Chiburis et al. (2011) provided some evidence based on Monte Carlo simulations and concluded that the choice of the model depends on different factors, an important one being the normality assumption of the error terms. They found that a violation of the normality assumption can lead to a severe bias in the bivariate probit estimates. Instead, Bhattacharya et al. (2006)'s simulations

suggest that bivariate probit is slightly more robust than IV to non-normality of the error terms. In this paper, the normality assumption is not satisfied, so we prefer to show the results from the bivariate probit model as a robustness check only.

When choosing between linear IV and 2SRI, Terza et al. (2008) argued that in cases of binary dependent variables, we should rely on nonlinear 2SRI as these are the only ones that produce consistent estimates of ATE. Basu et al. (2018) analysed the case of binary outcome, a binary treatment and binary instrument and found that the 2SLS method with binary IV produce consistent estimates of LATE, while 2SRI estimate of ATE and LATE are generally bias. Among 2SRI the type of residual form that minimises the bias for the ATE are the generalised residuals.

We compare the results obtained from these approaches as a sensitivity exercise to test whether (and how much) our results are affected by the empirical approach we adopt. We find that the marginal effects obtained with the 2SLS and the bivariate probit model do not differ substantially (Table 11). This can be interpreted as evidence that relying on a linear specification of the model is acceptable. On the contrary, when comparing the results obtained with 2SLS and 2SRI, we find that the estimates are very different. In particular, in the latter case, the estimated effect is 0.41, but it is not statistically significant.

6.2 Robustness check: baby's health measures

Gestational age at birth and birth weight may be endogenous because they are potentially affected by the mode of delivery. This may be the case if, for example, women with low birth weight babies and women at high risk of giving birth before their due date are more likely to receive intensive care before the birth of their child and are also more likely to end up with an unplanned caesarean. For this reason, as a sensitivity, we estimate linear IV models excluding these variables from the vector of covariates. As Table 12 shows, results do not change significantly compared to a model with the full set of regressors. This is not surprising as baby's health at birth (proxied by birth weight and gestational age) is not associated to mother's mental health (Table 7). However, because the delivery method may be correlated with the baby's health at birth, we prefer to include these controls in the models.²¹

²¹As a robustness check, we replace birth weight and gestational age with an alternative measure indicating whether the baby was hospitalised in the first 9 months after birth. In this case, we find that the coefficient is positive and statistically significant, suggesting that having a baby hospitalised increases the risk of postnatal depression (Table 12). However, an issue related to this measure, as opposed to birth weight (or other proxies measured at birth) is that it may be capturing the effect of the mother's mental health after childbirth. For example, Perry (2008) shows that psychologically distressed mothers tend to be less able to take care of their children, in particular, in terms of adherence to health regimes. If this is the case, baby hospitalisation after birth would not only capture the effect of the baby's health but also the mother's mental health after the delivery (and its effect on the motherãAŹs behaviour). For this reason, we prefer to include birth weight and gestational age in the model main specification.

6.3 Heterogeneity by mother's education

Women with greater resources or knowledge may be more capable of mitigating the negative events they experience. We test this assumption by splitting the sample into two groups of mothers: those with a university degree and those with a lower qualification (if any). When estimating the linear IV model for the two groups, we find no differences in the relationship between unplanned caesarean delivery and maternal mental health, although the coefficients become not significant (Table 13).

7 Discussion and Conclusions

This study contributes to the growing economic literature on the determinants and consequences of the increased utilisation of surgical procedures such as caesarean deliveries. It represents the first attempt to identify the causal effect of unplanned caesarean sections on the mothers' mental health, looking at whether this procedure is associated with an increase in the mothers' risk of developing postnatal depression in the first nine months after childbirth.

Results show that unplanned caesarean deliveries carry significant psychological risks, with women who give birth through this procedure being more vulnerable to depression (by 15 percentage points when estimated using linear IV models combined with hospital fixed effects). These findings are important for a number of reasons. First, caesarean deliveries have spread remarkably fast in recent years, becoming one of the most frequent surgical procedures, with 165,000 deliveries performed every year in England (among these, about 25,000 are unplanned caesarean deliveries). Second, depression can be a very severe condition, limiting the mothers' everyday life and their ability to take care of their children. Because mothers are usually the main childcare givers, the mother's poor mental health is likely to negatively affect their baby's health and development. Additionally, postnatal depression is likely to become a chronic health condition, associated with high costs for families as well as for society (e.g. inability to work, see (Schultz et al., 2013)).

This paper also explains the effect of caesarean deliveries on women's subsequent fertility decisions. Halla et al. (2016) find that mothers who give birth through a caesarean section are less likely to have other children and they mention psychological problems after childbirth as a possible explanation. While we cannot argue that this is necessarily the case, this paper shows evidence of the existence of a negative psychological impact of unplanned caesarean deliveries, which can explain their findings.

A limitation of this study concerns the data employed, more particularly, the limited information on the mother's mental health status during the pregnancy, together with the impossibility to distinguish between planned caesareans due to the mothers' or baby's health and those scheduled because of the mother's request. A longitudinal administrative dataset with detailed information on the mothers' previous pregnancy experiences and their health conditions before and after the pregnancy would allow for the identification of the causal effect of unplanned caesarean deliveries using alternative econometric approaches and fewer assumptions. In addition, it would allow the comparison of results from this study with those obtained using objective measures of the mother's mental health. Reaching these goals would require a link of hospital records on maternity events to other data sources containing information on primary care visits, being that depression is usually diagnosed by general practitioners (at least in a first instance), and census data providing details on mothers' income, education, working condition and other socio-economic variables. However, such linkages are not currently available, at least for English data.

The analysis of the effect of elective caesarean deliveries on the mother's mental health after childbirth is left to further research. As discussed in the paper, from a theoretical point of view, we may expect elective caesarean deliveries to have a smaller impact compared to unplanned caesarean deliveries, being planned in advance and giving mothers the opportunity to adjust their expectations. However, they can still have a negative impact on their body, and as a consequence, make their post-partum recovery more difficult. Looking also to elective caesareans would provide a more complete picture of this phenomenon.

From a policy perspective, this study highlights the importance of accounting for the psychological costs of unplanned caesarean deliveries when evaluating the costs and benefits of this procedure. Failing to account for these factors would lead to inaccurate evaluations of this procedure and, as a consequence, to the implementation of inappropriate health policies (Drummond, 2005). Additionally, it suggests the importance of providing appropriate services, such as professionally-based home visits and peer-based telephone support, to prevent the development of postnatal depression (Royal College of Obstetricians and Gynaecologists, 2017).

References

- Alexandersson, O., M. Bixo, and U. Högberg (2005). Evidence-based changes in term breech delivery practice in Sweden. Acta Obstetricia et Gynecologica Scandinavica 84(6), 584–587.
- Angrist, J. D. and J.-S. Pischke (2008). *Mostly harmless econometrics: An empiricist's companion*. Princeton university press.
- Basu, A., N. B. Coe, and C. G. Chapman (2018). 2sls versus 2sri: A ppropriate methods for rare outcomes and/or rare exposures. *Health Economics* 27(6), 937–955.
- Bhattacharya, J., D. Goldman, and D. McCaffrey (2006). Estimating probit models with self-selected treatments. *Statistics in Medicine* 25(3), 389–413.
- Boyce, P. M. and A. L. Todd (1992). Increased risk of postnatal depression after emergency. The Medical Journal of Australia 157, 172–174.
- Bragg, F., D. A. Cromwell, L. C. Edozien, I. Gurol-Urganci, T. A. Mahmood, A. Templeton, J. H. van der Meulen, et al. (2010). Variation in rates of caesarean section among English NHS trusts after accounting for maternal and clinical risk: cross sectional study. *British Medical Journal 341*.
- Brown, H. S. (1996). Physician demand for leisure: implications for cesarean section rates. Journal of Health Economics 15(2), 233–242.
- Canadian Mental Health Association (2008). The relationship between mental health, mental illness and chronic physical conditions. https://ontario.cmha.ca/public_policy/the-relationshipbetween-mental-health-mental-illness-and-chronic-physical-conditions/. [Online; accessed 25-November-2015].
- Card, D., A. Fenizia, and D. Silver (2018). The health effects of cesarean delivery for low-risk first births. Technical Report w24493, National Bureau of Economic Research.
- Carter, F. A., C. M. Frampton, and R. T. Mulder (2006). Cesarean section and postpartum depression: a review of the evidence examining the link. *Psychosomatic Medicine* 68(2), 321–330.
- Chiburis, R. C., J. Das, M. Lokshin, et al. (2011). A practical comparison of the bivariate probit and linear iv estimators. Technical Report 5601, The World Bank.
- Clement, S. (2001). Psychological aspects of caesarean section. Best Practice & Research Clinical Obstetrics & Gynaecology 15(1), 109–126.
- Coneus, K. and C. K. Spiess (2012). The intergenerational transmission of health in early childhood. Evidence from the German Socio-Economic Panel Stud. *Economics & Human Biology* 10(1), 89–97.

- Conger, K. J., M. A. Rueter, and R. D. Conger (2000). The role of economic pressure in the lives of parents and their adolescents: The Family Stress Model. In L. J. Crockett and R. K. Silbereisen (Eds.), *Negotiating adolescence in times of social change*. New York, NY, US: Cambridge University Press.
- Cooklin, A. R., L. H. Amir, J. Jarman, M. Cullinane, S. M. Donath, and C. S. Team (2015). Maternal physical health symptoms in the first 8 weeks postpartum among primiparous australian women. *Birth* 42(3), 254–260.
- Costa-Ramon, A. M., A. Rodriguez-Gonzalez, M. Serra-Burriel, and C. Campillo-Artero (2018). It's about time: Cesarean sections and neonatal health. *Journal of Health Economics* 59, 46 59.
- Deal, L. W. and V. L. Holt (1998). Young maternal age and depressive symptoms: results from the 1988 National Maternal and Infant Health Survey. American Journal of Public Health 88(2), 266–270.
- Declercq, E., F. Menacker, and M. MacDorman (2006). Maternal risk profiles and the primary cesarean rate in the united states, 1991–2002. *American Journal of Public Health* 96(5), 867.
- Drummond, M. F. (2005). *Methods for the economic evaluation of health care programmes*. Oxford University Press.
- Epstein, A. J. and S. Nicholson (2009). The formation and evolution of physician treatment styles: an application to cesarean sections. *Journal of Health Economics* 28(6), 1126–1140.
- Essex, H., J. Green, H. Baston, and K. Pickett (2013). Which women are at an increased risk of a caesarean section or an instrumental vaginal birth in the UK: an exploration within the Millennium Cohort Study. BJOG: An International Journal of Obstetrics & Gynaecology 120(6), 732–743.
- Fiscella, K., P. Franks, M. P. Doescher, and B. G. Saver (2002). Disparities in health care by race, ethnicity, and language among the insured: findings from a national sample. *Medical Care* 40(1), 52–59.
- Fisher, J., J. Astbury, and A. Smith (1997). Adverse psychological impact of operative obstetric interventions: a prospective longitudinal study. Australian and New Zealand Journal of Psychiatry 31(5), 728–738.
- Forman, D. N., P. Videbech, M. Hedegaard, J. D. Salvig, and N. Secher (2000). Postpartum depression: identification of women at risk. *British Journal of Obstetrics and Gynaecology* 107(10), 1210–1217.
- Fruscalzo, A., A. P. Londero, S. Salvador, S. Bertozzi, A. Biasioli, M. Della Martina, L. Driul, and D. Marchesoni (2014). New and old predictive factors for breech presentation: our experience in 14 433 singleton pregnancies and a literature review. *The Journal of Maternal-Fetal & Neonatal Medicine* 27(2), 167–172.

- Gavin, N. I., B. N. Gaynes, K. N. Lohr, S. Meltzer-Brody, G. Gartlehner, and T. Swinson (2005). Perinatal depression: a systematic review of prevalence and incidence. *Obstetrics & Gynecology* 106(5), 1071– 1083.
- Gholitabar, M., R. Ullman, D. James, M. Griffiths, et al. (2011). Caesarean section: summary of updated NICE guidance. *British Medical Journal 343*.
- Gibbons, L., J. M. Belizán, J. A. Lauer, A. P. Betrán, M. Merialdi, F. Althabe, et al. (2010). The global numbers and costs of additionally needed and unnecessary caesarean sections performed per year: overuse as a barrier to universal coverage. World Health Report 30, 1–31.
- Goffinet, F., M. Carayol, J.-M. Foidart, S. Alexander, S. Uzan, D. Subtil, G. Bréart, P. S. Group, et al. (2006). Is planned vaginal delivery for breech presentation at term still an option? results of an observational prospective survey in france and belgium. *American Journal of Obstetrics and Gynecology* 194(4), 1002–1011.
- Gresenz, C. R., R. Sturm, and L. Tang (2001). Income and mental health: unraveling community and individual level relationships. *Journal of Mental Health Policy and Economics* 4(4), 197–204.
- Grossman, M. (1972). On the concept of health capital and the demand for health. Journal of Political Economy, 223–255.
- Gruber, J. and M. Owings (1996). Physician financial incentives and cesarean section delivery. The Rand Journal of Economics 27(1), 99.
- Halla, M., H. Mayr, G. J. Pruckner, and P. García-Gómez (2016). Cutting fertility: The effect of cesarean deliveries on subsequent fertility and maternal labor supply. Working paper no. 14, University of Innsbruck Working Papers in Economics and Statistics.
- Hannah, M., W. Hannah, E. Hodnett, B. Chalmers, R. Kung, A. Willan, K. Amankwah, M. Cheng, M. Helewa, S. Hewson, et al. (2002). Term Breech Trial 3-Month Follow-up Collaborative Group. Outcomes at 3 months after planned cesarean vs planned vaginal delivery for breech presentation at term: the international randomized Term Breech Trial. JAMA: the Journal of the American Medical Association 287(14), 1822–31.
- Harris, K. M., M. J. Edlund, and S. Larson (2005). Racial and ethnic differences in the mental health problems and use of mental health care. *Medical Care* 43(8), 775–784.
- Health and Social Care Information Centre (2012). NHS Maternity Statistics 2011-12, Summary Report, Hospital Episode Statistics. http://www.hscic.gov.uk/catalogue/PUB09202/nhs-mate-eng-2011-2012-rep.pdf. [Online; accessed 25-November-2015].

- Heckman, J. J. (1978). Dummy endogenous variables in a simultaneous equation system. *Econometrica*, 931–959.
- Imbens, G. M. and J. M. Wooldridge (2008). Recent developments in the econometrics of program evaluation. Technical report, National Bureau of Economic Research.
- Impey, L. W. M., D. J. Murphy, M. GrifiňĄths, and L. Penna (2017, 6). Management of Breech Presentation. BJOG: An International Journal of Obstetrics Gynaecology 124(7), e151–e177.
- Jensen, V. M. and M. Wüst (2015). Can caesarean section improve child and maternal health? The case of breech babies. *Journal of Health Economics 39*, 289–302.
- Johnston, D. W., S. Schurer, and M. A. Shields (2013). Exploring the intergenerational persistence of mental health: Evidence from three generations. *Journal of Health Economics* 32(6), 1077–1089.
- Kiernan, K. and F. Mensah (2009). Maternal indicators in pregnancy and childrens infancy that signal future outcomes for childrens development, behaviour and health: evidence from the Millennium Cohort Study. York: University of York.
- Koechlin, F., L. Lorenzoni, and P. Schreyer (2010). Comparing price levels of hospital services across countries: Results of pilot study. Technical Report 53.
- Koo, V., J. Lynch, and S. Cooper (2003). Risk of postnatal depression after emergency delivery. Journal of Obstetrics and Gynaecology Research 29(4), 246–250.
- Lee, H. C., Y. Y. El-Sayed, and J. B. Gould (2008). Population trends in cesarean delivery for breech presentation in the United States, 1997-2003. American journal of obstetrics and gynecology 199(1), 59-e1.
- Lobel, M. and R. S. DeLuca (2007). Psychosocial sequelae of cesarean delivery: review and analysis of their causes and implications. *Social Science & Medicine* 64(11), 2272–2284.
- Lydon-Rochelle, M. T., V. L. Holt, and D. P. Martin (2001). Delivery method and self-reported postpartum general health status among primiparous women. *Paediatric and Perinatal Epidemiology* 15(3), 232–240.
- Macfarlane, A., B. Blondel, A. Mohangoo, M. Cuttini, J. Nijhuis, Z. Novak, H. Ólafsdóttir, and J. Zeitlin (2015). Wide differences in mode of delivery within europe: risk-stratified analyses of aggregated routine data from the euro-peristat study. BJOG: An International Journal of Obstetrics & Gynaecology.
- Mcintosh, J. (1993). The experience of motherhood and the development of depression in the postnatal period. *Journal of Clinical Nursing* 2(4), 243–249.

- McLennan, J. D., M. Kotelchuck, and H. Cho (2001). Prevalence, persistence, and correlates of depressive symptoms in a national sample of mothers of toddlers. *Journal of the American Academy of Child &* Adolescent Psychiatry 40(11), 1316–1323.
- Minkovitz, C. S., D. Strobino, D. Scharfstein, W. Hou, T. Miller, K. B. Mistry, and K. Swartz (2005). Maternal depressive symptoms and children's receipt of health care in the first 3 years of life. *Pedi-atrics* 115(2), 306–314.
- Mutryn, C. S. (1993). Psychosocial impact of cesarean section on the family: a literature review. Social Science & Medicine 37(10), 1271–1281.
- Nichols, A. (2011). Causal inference for binary regression. In *Stata conference Chicago*, Volume 14, pp. 2011.
- OECD (2015). Health at a glance 2015: OECD indicators. OECD publishing.
- OECD (2017). Health at a glance 2017: OECD indicators. OECD publishing.
- Patel, R. R., D. J. Murphy, T. J. Peters, et al. (2005). Operative delivery and postnatal depression: a cohort study. *British Medical Journal 330*(7496), 879.
- Patel, R. R., T. J. Peters, D. J. Murphy, et al. (2005). Prenatal risk factors for caesarean section. analyses of the alspac cohort of 12 944 women in England. *International Journal of Epidemiology* 34(2), 353–367.
- Perry, C. D. (2008). Does treating maternal depression improve child health management? The case of pediatric asthma. *Journal of Health Economics* 27(1), 157–173.
- Petrou, S., P. Cooper, L. Murray, and L. L. Davidson (2002). Economic costs of post-natal depression in a high-risk British cohort. The British Journal of Psychiatry 181(6), 505–512.
- Plewis, I., L. Calderwood, D. Hawkes, G. Hughes, and H. Joshi (2007). Millennium Cohort Study: technical report on sampling. *London: Centre for Longitudinal Studies*.
- Propper, C., J. Rigg, and S. Burgess (2007). Child health: evidence on the roles of family income and maternal mental health from a UK birth cohort. *Health Economics* 16(11), 1245–1269.
- Rayl, J., P. J. Gibson, and D. E. Hickok (1996). A population-based case-control study of risk factors for breech presentation. American Journal of Obstetrics and Gynecology 174(1), 28–32.
- Rietberg, C. C. T., P. M. Elferink-Stinkens, and G. H. Visser (2005). The effect of the Term Breech Trial on medical intervention behaviour and neonatal outcome in The Netherlands: an analysis of 35,453 term breech infants. BJOG: An International Journal of Obstetrics & Gynaecology 112(2), 205–209.

- Robertson, E., S. Grace, T. Wallington, and D. E. Stewart (2004). Antenatal risk factors for postpartum depression: a synthesis of recent literature. *General Hospital Psychiatry* 26(4), 289–295.
- Rosenzweig, M. R. and T. P. Schultz (1983). Estimating a household production function: Heterogeneity, the demand for health inputs, and their effects on birth weight. *The Journal of Political Economy*, 723–746.
- Royal College of Obstetricians and Gynaecologists (2006). The management of breech position. Guideline no. 20b. https://www.rcog.org.uk/globalassets/documents/guidelines/gtg-no-20bbreech-presentation.pdf. [Online; accessed 28-November-2015].
- Royal College of Obstetricians and Gynaecologists (2017). Maternal mental health: Women's voices. Technical report.
- Scheffler, R., N. Petris, F. Borgonovi, T. Brown, N. Petris, F. Sassi, P. Dourgnon, and N. Sirven (2010). Social capital, human capital and health: What is the evidence. http://www.oecd.org/edu/research/ 45760738.pdf.
- Schultz, D., K. A. Reynolds, L. M. Sontag-Padilla, S. L. Lovejoy, and R. Firth (2013). Transforming Systems for Parental Depression and Early Childhood Developmental Delays: Findings and Lessons Learned from the Helping Families Raise Healthy Children Initiative. Rand Corporation.
- Segre, L. S., M. W. O'Hara, S. Arndt, and S. Stuart (2007). The prevalence of postpartum depression. Social Psychiatry and Psychiatric Epidemiology 42(4), 316–321.
- Soet, J. E., G. A. Brack, and C. DiIorio (2003). Prevalence and predictors of women's experience of psychological trauma during childbirth. *Birth* 30(1), 36–46.
- Stewart, D. E., E. Robertson, C.-L. Dennis, S. L. Grace, and T. Wallington (2003). Postpartum depression: Literature review of risk factors and interventions. *Toronto: University Health Network Womens Health Program for Toronto Public Health*.
- Stock, J. H. and M. Yogo (2005). Testing for weak instruments in linear IV regression. *Identification and inference for econometric models: Essays in honor of Thomas Rothenberg.*
- Stuart-Parrigon, K. and S. Stuart (2014). Perinatal depression: An update and overview. Current Psychiatry Reports 16(9), 468.
- Terza, J. V., A. Basu, and P. J. Rathouz (2008). Two-stage residual inclusion estimation: addressing endogeneity in health econometric modeling. *Journal of Health Economics* 27(3), 531–543.

- Tharin, J. E., S. Rasmussen, and L. Krebs (2011). Consequences of the term breech trial in Denmark. Acta Obstetricia et Gynecologica Scandinavica 90(7), 767–771.
- Warner, R., L. Appleby, A. Whitton, and B. Faragher (1996). Demographic and obstetric risk factors for postnatal psychiatric morbidity. *The British Journal of Psychiatry* 168(5), 607–611.
- WHO (2015). WHO statement on caesarean section rates. Technical report, World Health Organisation, Geneva.
- Whyte, H., M. E. Hannah, S. Saigal, W. J. Hannah, S. Hewson, K. Amankwah, M. Cheng, A. Gafni, P. Guselle, M. Helewa, et al. (2004). Outcomes of children at 2 years after planned cesarean birth versus planned vaginal birth for breech presentation at term: the international randomized term breech trial. *American Journal of Obstetrics and Gynecology 191*(3), 864–871.

	Mean	Std. Dev.	Min	Max
Breech position	0.070	0.266	0	1
Postpartum depression	0.325	0.468	0	1
Mother's age	25.936	5.913	14	48
Married	0.493	0.500	0	1
Household income	16.979	10.919	1.21	66.70
GCSE/O-level (or equivalent)	0.386	0.487	0	1
A-level or higher (but no university)	0.178	0.382	0	1
University qualification	0.335	0.472	0	1
White	0.889	0.314	0	1
Black	0.024	0.153	0	1
Pakistani, Bangladeshi, Indian	0.059	0.236	0	1
Other ethinicity	0.028	0.164	0	1
Hospitalisation (pregnancy)	0.192	0.394	0	1
BMI (before pregnancy)	23.174	4.223	12.1	47.7
Planned pregnancy	0.528	0.499	0	1
Employed (pregnancy)	0.803	0.397	0	1
Diabetes (ever)	0.014	0.116	0	1
Hypertension (ever)	0.003	0.052	0	1
Kidney disease (ever)	0.009	0.095	0	1
Pre-eclampia (pregnancy)	0.089	0.285	0	1
Birth weight	3.372	0.474	0.67	4.89
Gestational age at birth	40.139	1.306	37	42

 Table 1: Descriptive Statistics

Notes. Mother's age is measured in years, while baby's gestational age at birth is measured in weeks. Income is measured in thousands of GB pounds. Statistics obtained using a sample of 5,896 first-time mothers. *** p<0.01, ** p<0.05 and * p<0.10.

	Natural delivery	Unplanned CS	P-value
Mother's age	25.43	28.09	0.000
Married	0.476	0.567	0.000
Household income	16.52	18.91	0.000
GCSE/O-level (or equivalent)	0.391	0.362	0.077
A-level or higher (but no university)	0.181	0.166	0.231
University qualification	0.324	0.383	0.000
White	0.891	0.882	0.384
Black	0.021	0.038	0.001
Pakistani, Bangladeshi, Indian	0.060	0.055	0.513
Other ethinicity	0.028	0.025	0.538
Hospitalisation (pregnancy)	0.188	0.210	0.092
BMI (before pregnancy)	22.87	24.46	0.000
Planned pregnancy	0.513	0.590	0.000
Employed (pregnancy)	0.792	0.850	0.000
Diabetes (ever)	0.011	0.023	0.002
Hypertension (ever)	0.002	0.007	0.002
Kidney disease (ever)	0.010	0.006	0.253
Pre-eclampia (pregnancy)	0.080	0.127	0.000
Birth weight	3.347	3.476	0.000
Gestational age at birth	40.10	40.29	0.000

Table 2: Mother's and baby's characteristics by delivery mode

Notes. The table reports the mean of the variables identifying mother's and baby's characteristics by delivery mode. Mother's age is measured in years, while baby's gestational age at birth is measured in weeks. Income is measured in thousands of GB pounds. All the other figures indicate percentages. Differences between groups are tested by means of 2-independent-sample t tests. Statistics obtained using a sample of 5,896 first-time mothers. *** p<0.01, ** p<0.05 and * p<0.10.

	Outcome variable: Breech position
Age	0.003
-	(0.005)
Age sq.	-0.000
01	(0.000)
Married	-0.014
	(0.009)
Income	0.000
Income	(0.000)
GCSE/O-level or eq.	-0.006
	(0.013)
A level or more (no uni)	-0.002
in level of more (no uni)	(0.013)
University qualification	0.002
University quantication	(0.013)
Black	-0.010
Diack	(0.021)
Debisteri Develedeshi Indian	· · · · ·
Pakistani, Bangladeshi, Indian	-0.014
	(0.013)
Other ethnicity (Asian)	-0.024
	(0.020)
Hospitalisations during pregnancy	0.019*
	(0.011)
BMI	0.017***
	(0.004)
BMI sq.	-0.000***
	(0.000)
Planned pregnancy	0.002
	(0.009)
Employed during pregnancy	0.006
	(0.009)
Diabetes	-0.022
	(0.028)
Hypertension	0.091
	(0.094)
Kidney disease	0.025
	(0.043)
Eclampsia	-0.015
	(0.014)
Under weight (b)	0.001
	(0.021)
Over weight (b)	0.018
	(0.013)
Gestational age (38)	0.005
	(0.020)
Gestational age (39)	0.005
Gestational age (55)	(0.020)
Gestational age (40)	-0.006
Gestational age (40)	-0.000 (0.018)
Gestational age (41)	-0.006
Gestational age (41)	
Contational $s = -(42)$	(0.017)
Gestational age (42)	-0.023
	(0.020)
Constant	-0.218**
	(0.085)
	()

Table 3: Validity test: Predictors of breech position

 $\begin{array}{c} \hline & 31 \\ \hline \textbf{Notes.} & \text{Robust standard errors clustered at hospital level in} \\ parentheses. & \text{Linear probability models estimated using OLS.} \\ ^{***} p{<}0.01, \ ^{**} p{<}0.05 \ \text{and} \ ^* p{<}0.10. \end{array}$

	No Breech	Breech	P-value
Age (m)	25.85	27.03	0.000
Married	0.492	0.508	0.52
Income	16.86	18.46	0.003
GCSE/O-level or eq.	0.388	0.357	0.195
A level or more (no uni)	0.178	0.175	0.874
University Qualification	0.331	0.386	0.018
Black	0.0242	0.0222	0.783
Pakistani, Bangladeshi, Indian	0.0612	0.0377	0.043
Hospitalisation	0.19	0.222	0.099
BMI	23.14	23.58	0.032
Planned pregnancy	0.524	0.57	0.063
Employed during pregnancy	0.799	0.854	0.005
Diabetes	0.0138	0.0111	0.635
Hypertension	0.0024	0.0067	0.094
Kidney diseases	0.0088	0.0133	0.336
Eclampsia	0.0885	0.0931	0.741
Birth weight	3.369	3.407	0.103
Gestational age (weeks)	40.15	40.05	0.135

Table 4: Validy test: Mean comparison

Notes. The table reports the mean of the variables employed in the analysis by position of the baby in the womb before birth (breech and normal position). Mother's age is measured in years, while baby's gestational age at birth is measured in weeks. Income is measured in thousands of GB pounds. All the other figures indicate percentages. Differences between groups are tested by means of 2-independent-sample t tests. *** p<0.01, ** p<0.05 and * p<0.10.

	Outcome Variable: Residuals from Breech Model
N. cigarettes	0.001
	(0.001)
Smoker (dummy)	-0.011
	(0.012)
Constant	0.001
	(0.004)
No. Observations	5,896
F statistic	0.49

Table 5: Validity test: Analysis of the unobservables

Notes. Robust standard errors clustered at hospital level in parentheses. Linear probability models estimated using OLS. The residuals employed as dependent variable are obtained from a model that predicts the probability of having a baby in a breech position on the covariates included in the main model (Table 3). Mother's smoking behavior refers to the period before the pregnancy. F statistic of joint significance of the independent variables. *** p<0.01, ** p<0.05 and * p<0.10.

Table 6: Linea	r probability	(LPM)) and hospital FE mode	els

		utcomo Var	inhle: Postn	atal Depression
	LPM	LPM	LPM	LPM with hospital FE
	(1)	(2)	(3)	(4)
Unplanned CS	0.029**	0.025*	0.025*	0.026*
	(0.014)	(0.015)	(0.015)	(0.015)
Mother's age	-0.013	-0.001	-0.002	0.001
	(0.009)	(0.009)	(0.009)	(0.009)
Mother's age sq.	0.000	0.000	0.000	0.000
	(0.000)	(0.000)	(0.000)	(0.000)
Married	-0.069***	-0.052***	-0.052***	-0.050***
	(0.015)	(0.016)	(0.016)	(0.016)
Household Income	-0.003***	-0.003***	-0.003***	-0.002***
	(0.001)	(0.001)	(0.001)	(0.001)
Black	0.080	0.075	0.074	0.062
	(0.050)	(0.050)	(0.050)	(0.057)
Pakistani, Bangladeshi, Indian	0.085***	0.079***	0.077***	0.056*
, , , , , , , , , , , , , , , , , , , ,	(0.023)	(0.023)	(0.023)	(0.030)
Other ethnicity	-0.016	-0.025	-0.027	-0.018
2	(0.039)	(0.039)	(0.040)	(0.038)
GCSE/O-level (or equivalent)	0.013	0.022	0.022	0.016
	(0.013)	(0.022)	(0.022)	(0.021)
A-level or higher (but no university)	-0.009	0.006	0.006	-0.004
reference in fighter (but no university)	(0.026)	(0.025)	(0.025)	(0.024)
University qualification	-0.013	-0.001	-0.002	-0.010
University quantication	(0.026)	(0.026)	(0.026)	(0.024)
BMI	(0.020)	0.003	0.003	0.002
Ditt		(0.010)	(0.010)	(0.010)
BMI sq.		-0.000	-0.000	-0.000
Dim sq.		(0.000)	(0.000)	(0.000)
Planned pregnancy		-0.056***	-0.056***	-0.062***
ranned pregnancy		(0.013)	(0.013)	(0.014)
Hospitalisation (pregnancy)		0.111***	0.112***	0.112***
nospitalisation (pregnancy)		(0.017)	(0.017)	(0.018)
Employed (pregnancy)		-0.048**	-0.048**	-0.043**
Employed (pregnancy)		(0.019)	(0.040)	(0.020)
Diabetes (ever)		-0.033	-0.031	-0.028
Diabetes (ever)		(0.050)	(0.051)	(0.054)
Hypertension (ever)		-0.039	-0.037	-0.039
Hypertension (ever)		(0.115)		
Kidney diseases (ever)		(0.115) 0.059	(0.115) 0.061	(0.127) 0.060
Kidney diseases (ever)				
Unweight baby		(0.062)	(0.062)	(0.064)
Unweight baby			0.017	0.019
Overweight baby			(0.035) -0.006	(0.036)
Overweight baby				-0.007
28th woold (most at icm)			(0.019)	(0.020)
38th week (gestation)			0.004	0.009
			(0.037)	(0.039)
39th week (gestation)			0.040	0.037
			(0.037)	(0.039)
40th week (gestation)			0.022	0.016
			(0.033)	(0.035)
41th week (gestation)			0.017	0.014
			(0.035)	(0.036)
42th week or more (gestation)			0.041	0.040
_			(0.033)	(0.035)
Constant	0.563***	0.367**	0.345**	0.326**
	(0.115)	(0.147)	(0.147)	(0.144)
No. Observations	5,896	5,896	5,896	5,896

 $\begin{array}{c} 34\\ \hline \textbf{Notes.} \ \text{Robust standard errors clustered at hospital level in parentheses.} \\ \ \text{Linear probability models estimated using OLS. Column 4 reports results} \\ \ \text{obtained including hospital fixed effects (FE). *** p<0.01, ** p<0.05 and $$ * p<0.10.$ \\ \end{array}$

	Linear IV m	odel without FE	Linear IV n	nodel with FE
	First stage	Second stage	First stage	Second stag
	(1)	(2)	(3)	(4)
Breech baby	0.310***		0.314***	
Unplanned CS	(0.024)	0.188***	(0.024)	0.151**
Mother's age	-0.006	(0.069) -0.001	-0.005	(0.069) 0.002
Mother's age sq.	(0.008)	(0.009)	(0.008)	(0.009)
	0.000^{**}	0.000	0.000^{**}	0.000
Married	(0.000)	(0.000)	(0.000)	(0.000)
	0.001	-0.051***	-0.005	-0.049***
Household Income	(0.013)	(0.016)	(0.013)	(0.016)
	0.000	-0.003***	0.000	-0.002***
Black	(0.001)	(0.001)	(0.001)	(0.001)
	0.106^{***}	0.057	0.108**	0.049
Pakistani, Bangladeshi, Indian	(0.040) 0.046^{**}	(0.054) 0.070^{***}	(0.046) 0.028 (0.028)	(0.060) 0.054^{*}
Other ethnicity	(0.023)	(0.023)	(0.028)	(0.030)
	0.006	-0.027	0.010	-0.019
	(0.028)	(0.040)	(0.033)	(0.038)
GCSE/O-level (or equivalent)	(0.028)	(0.040)	(0.033)	(0.038)
	-0.026	0.027	-0.024	0.019
	(0.017)	(0.023)	(0.017)	(0.022)
A-level or higher (but no university)	-0.043^{**} (0.019)	(0.026) (0.026)	-0.037^{*} (0.019)	(0.022) 0.001 (0.025)
University qualification	-0.051** (0.020)	0.006 (0.026)	-0.046** (0.021)	-0.004 (0.025)
BMI	-0.009 (0.008)	0.003 (0.010)	-0.006 (0.008)	0.002 (0.010)
BMI sq.	0.000^{**}	-0.000	0.000^{**}	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)
Planned pregnancy	-0.014	-0.053^{***}	-0.011	-0.061^{***}
	(0.013)	(0.014)	(0.013)	(0.014)
Hospitalisation (pregnancy)	0.023^{*}	0.107^{***}	0.023^{*}	0.109^{***}
	(0.013)	(0.018)	(0.013)	(0.018)
Employed (pregnancy)	0.005	-0.049^{**}	0.011	-0.044^{**}
	(0.013)	(0.019)	(0.014)	(0.020)
Diabetes (ever)	0.089^{*}	-0.045	0.116^{**}	-0.040
	(0.049)	(0.051)	(0.051)	(0.054)
Hypertension (ever)	0.201	-0.074	0.188	-0.066
	(0.127)	(0.112)	(0.132)	(0.126)
Kidney diseases (ever)	-0.066 (0.047) 0.079^{***}	0.071 (0.062)	-0.068 (0.049)	0.067 (0.063) 0.008
Unweight baby Overweight baby	(0.029) 0.145^{***}	0.004 (0.037) -0.031	0.086^{***} (0.029) 0.151^{***}	(0.008 (0.037) -0.026
38th week (gestation)	(0.019)	(0.020)	(0.020)	(0.021)
	-0.016	0.007	-0.019	0.011
39th week (gestation)	(0.030)	(0.038)	(0.031)	(0.039)
	-0.019	0.043	-0.028	0.041
40th week (gestation)	(0.026)	(0.037)	(0.027)	(0.039)
	-0.019	0.025	-0.022	0.019
11th week (gestation)	(0.027)	(0.033)	(0.028)	(0.035)
	0.018	0.014	0.016	0.013
42th week or more (gestation)	(0.026)	(0.034)	(0.027)	(0.036)
	0.046	0.035	0.043	0.035
Constant	(0.028) 0.091 (0.136)	(0.033) 0.341^{**} (0.145)	(0.029)	(0.036)
No. Obcommiti	. ,	. ,	-	961
No. Observations F-stat (first stage)		5,896 163.6		,864 67.3
,				
Weak identification test	103	.6;16.38	107.3	3;16.38

 $\frac{\text{Endogeneity Test (p-value)}}{35}$ **Notes.** Robust standard errors clustered at hospital level in parentheses. For the weak identification test we report Stock-Yogo critical values, allowing for 10% of IV maximum distortion with respect to OLS. *** p<0.01, ** p<0.05 and * p<0.10.

0.0134

0.0583

Endogeneity Test (p-value)

	Outcome Variable: Postnatal Depression
Unplanned CS	0.042**
	(0.019)
Mother's age	0.003
	(0.011)
Mother's age sq.	0.000
	(0.000)
Married	-0.039**
	(0.020)
Household Income	-0.002*
	(0.001)
Black	0.087
	(0.056)
Pakistani, Bangladeshi, Indian	0.033
, , ,	(0.039)
Other ethnicity	0.021
	(0.042)
GCSE/O-level (or equivalent)	0.013
	(0.024)
A-level or higher (but no university)	0.003
	(0.030)
University qualification	-0.013
Chivership qualification	(0.029)
BMI	-0.003
DIVIL	(0.011)
BMI sq.	0.000
DMI 5q.	(0.000)
Planned pregnancy	-0.080***
I failled pregnancy	(0.015)
Employed (pregnancy)	-0.054**
Employed (pregnancy)	(0.022)
Unweight haby	0.056
Unweight baby	
Overweight baby	(0.045) -0.022
Overweight baby	
38th week (gestation)	(0.023)
Sour week (gestation)	-0.017
20th much (martation)	(0.053)
39th week (gestation)	0.007
(0) the second (second a time)	(0.044)
40th week (gestation)	-0.032
(1+1)	(0.043)
41th week (gestation)	-0.028
49th mode on the first (the)	(0.043)
42th week or more (gestation)	0.006
	(0.045)
Constant	0.377**
	(0.172)
No. Observations	4,205

 $\begin{array}{c} 36\\ \hline \textbf{Notes.} \ \text{Robust standard errors clustered at hospital level in parentheses.} \\ The sample includes mother who did not suffered from diabetes, hypertension, kidney diseases or pre-eclampsia. *** p<0.01, ** p<0.05 and * p<0.10. \\ \end{array}$

	(1)	Outcome Variable: Unplanned CS (2)
Midnight	(-)	0.040
mangni		(0.040) (0.037)
1am		-0.038
10111		(0.038)
2am		-0.015
20111		(0.030)
3am		-0.062**
Jaiil		(0.031)
4am		-0.073**
таш		(0.031)
5am		(0.031) -0.095***
Jaill		
6am		(0.030) -0.050
Valli		
7am		(0.033) -0.022
18111		
Pam		(0.035)
8am		-0.040
0.0.000		(0.033)
9am		-0.055
10		(0.033)
10am		-0.031
11		(0.034)
11am		0.005
		(0.032)
midday		-0.045
4		(0.030)
1pm		-0.040
9.222		(0.036)
2pm		-0.041
9		(0.033)
3pm		-0.036
4		(0.033)
4pm		-0.068**
F		(0.034)
5pm		-0.009
C		(0.033)
6pm		-0.077**
7.000		(0.031)
7pm		-0.059*
Q		(0.032)
8pm		-0.017
0		(0.034)
9pm		-0.019
10		(0.034)
10pm		-0.008
	0.000	(0.032)
hour of the day	0.000	
	(0.001)	
No. Observations	5,896	$5,\!896$
F statistic (first stag	e) 0.32	2.41

Table 9: First stage of linear IV models using time of the delivery as IV

Notes. Robust standard errors clustered at hospital level in parentheses. All model specifications include information on socio-economic status, pregnancy, mother's health and baby's health as in Column (3) of Table 6. Time is defined as a continuous measure (Column 1) and as a set of dummies (Column 2) capturing the time in a day when the baby was born. F statistic of the joint significance of the exclusion restrictions. *** p<0.01, ** p<0.05 and * p<0.10.

Table 10: Effect of unplanned CS on mother's mental health for first-time and multiple-time mothers

(1) (2) (3) (4) Breech baby 0.314*** 0.317*** (0.20) Unplanned CS 0.180*** 0.110*** (0.020) Mother's age 0.009** 0.002 0.009** 0.00 Mother's age sq. -0.000 -0.000 -0.000 -0.001 -0.001 Morther's age sq. -0.001 -0.004 -0.003 (0.000) (0.000) Household Income 0.001 -0.004 -0.003 (0.000) (0.000) Black 0.075*** -0.023 0.062*** -0.013 Pakistani, Banghadeshi, Indian 0.012 (0.017) (0.012) GCSE/O-level (or equivalent) -0.015 0.001 -0.013 0.000 GCSE/O-level (or equivalent) -0.015 0.003 -0.013 0.000 BMI sq. 0.000 0.000 0.000 0.000 0.000 BMI sq. 0.001 -0.013 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000		Linear IV m	odel without FE	Linear IV r	nodel with FI
Breech baby 0.14*** 0.13*** 0.140** Unplanned CS 0.160*** 0.050 Mother's age 0.0004 0.0006 0.0004 Mother's age sq. -0.000 -0.000 -0.000 Mother's age sq. -0.000 -0.000 -0.000 Mother's age sq. -0.001 -0.004 -0.007 Mother's age sq. -0.001 -0.007 -0.001 -0.007 Household Income 0.001 -0.003 0.001 -0.003 Back 0.075*** -0.023 0.062*** -0.013 Pakistani, Bangladeshi, Indian 0.012 -0.013 0.011 -0.033 CSE/O-level (or equivalent) -0.015 0.000 0.000 0.001 -0.015 0.000 CSE/O-level (or equivalent) -0.015 0.000 0.000 0.000 0.000 0.000 BMI 0.004 -0.005 0.006 0.000 0.000 0.000 BMI 0.000 0.0000 0.0000 0.0001 -0.014		First stage	Second stage	First stage	Second stag
(0.020) (0.020) Unplanned CS 0.009** 0.002 0.009** 0.000 Mother's age 0.009** 0.002 0.009** 0.000 Mother's age sq. 0.000 0.0000 0.0000 0.000 Mother's age sq. 0.001 -0.004 -0.005*** -0.001 Household Income 0.001 -0.004*** 0.001 -0.003 Black 0.007** -0.023 0.062*** -0.03 GCSE/O-level (or equivalent) 0.019 -0.05** 0.007 -0.03 GCSE/O-level (or equivalent) -0.015 0.000 0.000 0.000 COUTO* 0.001 0.001 0.001 0.001 0.001 Lawer on higher (but no university) -0.015 0.000 0.000 0.000 COUTO 0.0001 0.0001 0.0001 0.0001 0.000 Lawer on higher (but no university) -0.015 0.000 0.000 0.000 0.000 0.000 0.000 0.0001 0.000 0.000		(1)	(2)	(3)	(4)
Unplanned CS 0.180*** 0.146* Mother's age 0.00** 0.000 0.000 Mother's age sq. -0.000 -0.000 -0.000 -0.000 Mother's age sq. -0.001 -0.000 -0.000 -0.000 Married -0.004 -0.056*** -0.004 -0.057** Household Income 0.001 -0.004*** 0.001 -0.001 Back 0.075*** -0.023 0.062*** -0.013 Pakistani, Bangladeshi, Indian 0.018* -0.013 0.011 -0.002 Other ethnicity 0.019 -0.055* 0.007 -0.023 Pakistani, Bangladeshi, Indian 0.017* (0.010) (0.010) (0.010) Other ethnicity 0.019 -0.015* 0.000 0.000 0.000 Ciste/O-level (or equivalent) -0.018* 0.003 -0.011 0.001 University qualification 0.000 0.000 0.000 0.000 BM 0.0001 0.0000 0.0001 0.000	Breech baby				
Mother's age 0.009** 0.002 0.009** 0.00 Mother's age sq. -0.000 -0.000 -0.000 -0.000 Married -0.004 -0.055*** -0.004 -0.007 Household Income 0.001 -0.004*** 0.001 -0.004 Household Income 0.001 -0.004*** 0.001 -0.003 Black 0.075*** -0.023 0.062*** -0.013 Pakistani, Bangladeshi, Indian 0.018 (0.017) (0.012) (0.022) Other ethnicity 0.019 -0.056* 0.007 -0.013 Other ethnicity 0.019 -0.055 0.016 -0.002 0.012 COSE/O-level (or equivalent) -0.015 0.003 -0.013 0.000 University qualification -0.015 0.003 -0.013 0.000 BMI 0.000 0.000 0.000 0.000 0.000 Fist child (parity) 0.13*** -0.041*** -0.041 -0.044** 0.000 0.000	Unplanned CS	(0.020)		(0.020)	0.146***
Mother's age sq. -0.000 -0.000 -0.000 -0.000 -0.000 Married -0.004 $-0.05e^{334}$ -0.004 $-0.05e^{334}$ -0.004 Household Income 0.001 -0.004^{***} 0.001 -0.004^{***} 0.001 Black 0.075^{***} -0.023 0.062^{***} -0.013 Pakistani, Bangladeshi, Indian 0.019 -0.033 0.017 -0.033 Other ethnicity 0.019 -0.050^{**} 0.007 -0.03 CGSE/O-level (or equivalent) -0.015 0.002 0.012 0.007 0.001 A-level on higher (but no university) -0.015 0.003 -0.015 0.000 0.000 0.000 Uriversity qualification -0.015 0.003 -0.013 0.000 0.000 0.000 BMI 0.005 0.006 0.000 0.000 0.000 0.000 Fist child (parity) 0.022^{**} 0.091^{**} 0.002^{**} 0.002^{**}	Mother's age		0.002		(0.055) 0.004
Married -0.004 -0.0056*** -0.004 -0.007 Household Income 0.001 -0.004*** 0.001 -0.001 Black 0.075*** -0.023 0.062*** -0.013 Pakistani, Bangladeshi, Indian 0.019 -0.013 0.011 -0.002 Ohne ethnicity 0.019 -0.050* 0.007 -0.023 Odder ethnicity 0.019 -0.050* 0.007 -0.023 Ocder ethnicity 0.019 -0.050* 0.007 -0.013 Aclevel or higher (but no university) -0.015* 0.000 -0.013 0.001 University qualification -0.015 0.003 -0.013 0.001 University qualification -0.015 0.000 0.000 0.000 BM 0.000 0.000 0.000 0.000 0.000 BM on 0.000 0.000 0.000 0.000 0.000 BM on 0.001 -0.044*** -0.011 -0.044 BM on 0.0003 -0.044***<	Mother's age sq.	-0.000	-0.000	-0.000	(0.006) -0.000
Honschold Income 0.001 -0.004*** 0.001 -0.003 Black 0.075*** -0.023 0.062*** -0.013 Pakistani, Bangladeshi, Indian 0.019 -0.013 0.001 -0.003 Pakistani, Bangladeshi, Indian 0.019* -0.013 0.001 -0.002 Other ethnicity 0.002 (0.020) (0.025) (0.020) CGSE/O-level (or equivalent) -0.015 0.006 -0.010 (0.010) A-level or higher (but no university) -0.015* 0.008 -0.015 0.001 University qualification -0.015 0.003 -0.013 0.001 University qualification -0.015 0.006 (0.000) 0.000 BMI 0.0001 (0.0005) (0.0005 (0.000) 0.000 BMI sq. 0.0001 0.0001 0.0001 0.0001 0.0001 Fist child (parity) 0.135*** -0.046*** 0.031 -0.041** 0.021** 0.0894 Imployed (pregnancy) 0.022** 0.091*** 0.021** 0.021** 0.022** Imployed (pregnancy)	Married	. ,	. ,	. ,	(0.000) - 0.057^{***}
Black 0.075*** -0.023 0.062*** -0.01 Pakistani, Bangladeshi, Indian 0.019* -0.013 0.011 -0.02 Pakistani, Bangladeshi, Indian 0.019* -0.013 0.011 -0.02 Other ethnicity 0.019 -0.050* 0.007 -0.03 CGSE/O-level (or equivalent) -0.015 0.0007 (0.022) 0.019 -0.015 0.000 A-level or higher (but no university) -0.015 0.003 0.015 0.000 0.0010 (0.010) (0.010) (0.010) (0.010) 0.001 0.001 0.0000 0.0000 0.0000 <td>Household Income</td> <td></td> <td></td> <td></td> <td>(0.011) -0.003***</td>	Household Income				(0.011) -0.003***
Pakistani, Bangladeshi, Indian 0.019* -0.013 0.011 -0.003 Other ethnicity 0.019 -0.005* 0.007 -0.03 GCSE/O-level (or equivalent) -0.005 0.016 -0.002 0.012 GCSE/O-level (or equivalent) -0.015 0.016 -0.002 0.011 A-level or higher (but no university) -0.015 0.003 -0.013 0.001 University qualification -0.015 0.003 -0.013 0.001 University qualification -0.015 0.003 -0.013 0.001 BMI 0.000 0.000 0.000 0.000 0.000 BMI sq. 0.000 0.000 0.000 0.000 0.000 First child (parity) 0.135*** -0.046*** 0.035 -0.040 Ibaned pregnancy -0.001 -0.046*** 0.005 -0.040 Ibaned pregnancy 0.003 -0.046*** 0.005 -0.040 Ibaned pregnancy 0.003 -0.046*** 0.005 -0.040 <	Black	. ,		, ,	(0.001) -0.013
Other ethnicity 0.019 -0.050^{*} 0.007 -0.03 GCSE/O-level (or equivalent) -0.005 0.016 -0.002 0.014 A-level or higher (but no university) -0.018^{*} 0.008 -0.015 0.000 Mathematication -0.015 0.003 -0.013 0.000 University qualification -0.015 0.003 -0.013 0.000 BMI 0.004 -0.005 0.004 -0.005 BMI 0.000 0.000 0.000 0.000 0.000 BMI sq. 0.000 0.000 0.000 0.000 0.000 First child (parity) 0.135^{****} -0.046^{****} 0.031^{**} 0.021^{**} 0.041^{**} Planned pregnancy -0.001 -0.044^{****} -0.001 -0.046^{***} 0.005^{*} 0.004^{*} Ibapida (pregnancy) 0.022^{**} 0.001^{**} 0.005^{*} 0.001^{**} Ibapida (pregnancy) 0.005^{**} 0.004^{**} 0.005^{**}	Pakistani, Bangladeshi, Indian	. ,	. ,	. ,	(0.038) -0.002
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Other ethnicity		. ,	0.007	(0.023) -0.035
A-level or higher (but no university) -0.018^* 0.008 -0.015 0.000 University qualification -0.015 0.003 -0.013 0.001 University qualification -0.015 0.003 -0.013 0.001 BMI 0.004 -0.005 0.004 -0.005 BMI 0.004 -0.005 0.000 0.000 0.000 BMI sq. 0.000 0.000 0.000 0.000 0.000 Pist child (parity) 0.135^{***} -0.044^{****} 0.031^{*} -0.001 -0.044^{***} Planned pregnancy -0.001 -0.044^{***} -0.001 -0.044^{***} -0.001 -0.044^{***} Employed (pregnancy) 0.022^{**} 0.091^{***} 0.005 -0.040 Employed (pregnancy) 0.003 -0.044^{***} 0.005^{*} -0.042^{**} Unweight baby 0.057 0.044^{**} 0.005^{*} 0.044^{**} Unweight baby 0.092^{***} 0.032^{*} 0.041^{**} Unweight baby 0.002^{***} 0.016^{***}	GCSE/O-level (or equivalent)	-0.005	0.016	-0.002	(0.028) 0.017
$\begin{array}{llllllllllllllllllllllllllllllllllll$	A-level or higher (but no university)	. ,		. ,	(0.016) 0.007
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	University qualification	-0.015	0.003	-0.013	(0.016) 0.004
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	BMI		-0.005	0.004	(0.018) -0.005
First child (parity) 0.135^{***} -0.046^{***} 0.133^{***} -0.041 Planned pregnancy -0.001 -0.044^{***} -0.001 -0.044^{***} Planned pregnancy) 0.022^{**} 0.091^{***} 0.021^{**} 0.089^{**} Hospitalisation (pregnancy) 0.022^{**} 0.091^{***} 0.021^{**} 0.089^{**} Employed (pregnancy) 0.003 -0.040^{***} 0.005 -0.040 Diabetes (ever) 0.061^{**} -0.025 0.061^{**} -0.02 Hypertension (ever) 0.057 0.044 0.057 0.047 Unweight baby 0.092^{**} 0.032 0.091^{***} 0.032 Unweight baby 0.092^{**} 0.032 0.091^{***} 0.017 Mondel (gestation) -0.066^{***} -0.044^{***} 0.067^{***} -0.041 Unweight week (gestation) -0.064^{***} -0.041^{***} 0.007^{***} 0.011 Mondel (gestation) -0.066^{***} 0.003 -0.072^{***} 0.001 10.016 (0.022) (0.017) $(0.02$ <t< td=""><td>BMI sq.</td><td>0.000</td><td>0.000</td><td>0.000</td><td>(0.006) 0.000</td></t<>	BMI sq.	0.000	0.000	0.000	(0.006) 0.000
Planned pregnancy -0.001 -0.044^{***} -0.001 -0.044 (0.006) (0.008) (0.006) (0.009) Hospitalisation (pregnancy) 0.022^{**} 0.091^{***} 0.021^{**} 0.099 Employed (pregnancy) 0.003 -0.040^{***} 0.005 -0.040 Diabetes (ever) 0.061^{**} -0.025 0.061^{**} -0.026 Uppertension (ever) 0.057 0.044 0.057 0.044 Unweight baby 0.092^{**} 0.033 -0.005 0.007 Unweight baby 0.092^{**} 0.032 0.091^{***} 0.007 Unweight baby 0.092^{**} 0.032 0.091^{***} 0.007 Unweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041^{**} (0.021) (0.025) (0.016) (0.022) (0.019) (0.021) Overweight baby -0.066^{***} -0.040^{***} 0.067^{***} -0.013 (0.016) (0.024) (0.016) (0.022) (0.017) (0.022) (0.016) <	First child (parity)	0.135***	-0.046***	0.133***	(0.000) -0.041***
Hospitalisation (pregnancy) 0.022^{**} 0.091^{***} 0.021^{**} 0.089^4 Employed (pregnancy) 0.003 -0.040^{***} 0.005 -0.040 Diabetes (ever) 0.061^{**} -0.025 0.061^{**} -0.026 Diabetes (ever) 0.057 0.044 0.057 0.044 Diabetes (ever) 0.057 0.044 0.057 0.044 Hypertension (ever) 0.057 0.044 0.057 0.044 Unweight diseases (ever) -0.000 0.083^* -0.005 0.077 (0.021) (0.025) (0.021) (0.021) (0.020) (0.021) (0.020) Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 (0.009) (0.013) (0.009) (0.013) (0.009) (0.013) Sth week (gestation) -0.066^{***} -0.013 -0.057^{***} -0.010 (0.016) (0.024) (0.016) (0.022) (0.017) (0.021) (0.016) (0.022) (0.017) (0.022)	Planned pregnancy	-0.001	-0.044***	-0.001	(0.013) -0.044***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Hospitalisation (pregnancy)	0.022**	0.091***	0.021**	0.089***
Diabetes (ever) 0.061^{**} -0.025 0.061^{**} -0.025 Hypertension (ever) 0.057 0.044 0.057 0.044 Hypertension (ever) 0.057 0.044 0.057 0.044 Kidney diseases (ever) -0.000 0.083^* -0.005 0.071 Kidney diseases (ever) -0.000 0.083^* -0.005 0.077 (0.031) (0.045) (0.032) (0.047) (0.021) (0.020) Unweight baby 0.092^{***} 0.032 0.091^{***} -0.041 (0.021) (0.025) (0.021) (0.020) (0.019) (0.012) Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 (0.019) (0.023) (0.019) (0.020) (0.019) (0.020) 39th week (gestation) -0.066^{***} 0.003 -0.072^{***} 0.000 (0.016) (0.022) (0.017) (0.022) (0.017) (0.022)	Employed (pregnancy)	0.003	-0.040***	0.005	(0.011) -0.040***
Hypertension (ever) 0.057 0.044 0.057 0.044 (0.057) (0.074) (0.057) (0.07 Kidney diseases (ever) -0.000 0.083^* -0.005 0.077 (0.031) (0.045) (0.032) (0.044) Unweight baby 0.092^{***} 0.032 0.091^{***} 0.032 Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 Sth week (gestation) -0.046^{***} -0.041^{***} 0.009 (0.013) (0.009) (0.019) 39th week (gestation) -0.046^{***} 0.016 -0.072^{***} 0.010 (0.016) (0.024) (0.016) (0.022) (0.017) (0.022) 40th week (gestation) -0.052^{***} -0.002 -0.056^{***} -0.002 (0.017) (0.022) (0.017) (0.022) (0.017) (0.022) 42th week (gestation)	Diabetes (ever)	0.061**	-0.025	0.061**	-0.021
Kidney diseases (ever) -0.000 0.083^* -0.005 0.077 (0.031) (0.045) (0.032) (0.04 Unweight baby 0.092^{***} 0.032 0.091^{***} 0.032 Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 Sth week (gestation) -0.046^{***} -0.013 -0.050^{***} -0.01 38th week (gestation) -0.064^{***} 0.016 -0.072^{***} 0.01 (0.019) (0.023) (0.019) (0.02) 39th week (gestation) -0.066^{***} 0.003 -0.072^{***} 0.010 (0.016) (0.024) (0.016) (0.022) (0.017) (0.02) 40th week (gestation) -0.052^{***} -0.002 -0.056^{***} -0.002 (0.017) (0.022) (0.017) (0.02) (0.017) (0.02) 42th week or more (gestation) -0.025^{***} -0.006^{***} -0.030^{*} -0.00 <td< td=""><td>Hypertension (ever)</td><td>0.057</td><td>0.044</td><td>0.057</td><td>0.045</td></td<>	Hypertension (ever)	0.057	0.044	0.057	0.045
Unweight baby 0.092^{***} 0.032 0.091^{***} 0.032 Overweight baby (0.021) (0.025) (0.021) (0.020) Overweight baby 0.066^{***} -0.040^{***} 0.067^{***} -0.041 (0.009) (0.013) (0.009) (0.013) (0.009) (0.013) 38th week (gestation) -0.046^{***} -0.013 -0.050^{***} -0.01 39th week (gestation) -0.064^{***} 0.016 -0.072^{***} 0.01 40th week (gestation) -0.066^{***} 0.003 -0.072^{***} 0.00 40th week (gestation) -0.066^{***} 0.003 -0.072^{***} 0.00 40th week (gestation) -0.052^{***} -0.002 (0.017) (0.022) 41th week (gestation) -0.025^{***} -0.002 -0.030^{**} -0.002 42th week or more (gestation) -0.025^{***} -0.006 -0.030^{*} -0.002 Constant -0.243^{***} 0.502^{***} 0.002 0.017	Kidney diseases (ever)	-0.000	0.083*	-0.005	0.077*
$\begin{array}{c cccccc} Overweight baby & 0.066^{***} & -0.040^{***} & 0.067^{***} & -0.041 \\ (0.009) & (0.013) & (0.009) & (0.01) \\ 38th week (gestation) & -0.046^{***} & -0.013 & -0.050^{***} & -0.01 \\ (0.019) & (0.023) & (0.019) & (0.02) \\ 39th week (gestation) & -0.064^{***} & 0.016 & -0.072^{***} & 0.01 \\ (0.016) & (0.024) & (0.016) & (0.02) \\ 40th week (gestation) & -0.066^{***} & 0.003 & -0.070^{***} & 0.00 \\ (0.016) & (0.022) & (0.017) & (0.02) \\ 40th week (gestation) & -0.052^{***} & -0.002 & -0.056^{***} & -0.00 \\ (0.017) & (0.022) & (0.017) & (0.02) \\ 42th week or more (gestation) & -0.025 & -0.006 & -0.030^{*} & -0.00 \\ (0.017) & (0.023) & (0.017) & (0.02) \\ 42th week or more (gestation) & -0.025 & -0.006 & -0.030^{*} & -0.00 \\ (0.017) & (0.023) & (0.017) & (0.02) \\ Constant & -0.243^{***} & 0.502^{***} \\ (0.081) & (0.113) \\ \hline No. Observations & 13,368 & 13,340 \\ \hline F-stat & 254.1 & 262.4 \\ \hline \end{array}$	Unweight baby	0.092***	0.032	0.091***	0.034
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Overweight baby	0.066***	-0.040***	0.067***	-0.041***
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	38th week (gestation)	-0.046**	-0.013	-0.050***	-0.011
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	39th week (gestation)	-0.064***	0.016	-0.072***	0.016
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	40th week (gestation)	-0.066***	0.003	-0.070***	0.002
42th week or more (gestation) -0.025 -0.006 -0.030^* -0.00 (0.017) (0.023) (0.017) (0.02 Constant -0.243^{***} 0.502^{***} (0.017) (0.02 No. Observations 13,368 13,340 F-stat 254.1 262.4	41th week (gestation)	-0.052***	-0.002	-0.056***	-0.002
Constant -0.243*** 0.502*** (0.081) (0.113) No. Observations 13,368 13,340 F-stat 254.1 262.4	42th week or more (gestation)	-0.025	-0.006	-0.030*	-0.005
No. Observations 13,368 13,340 F-stat 254.1 262.4	Constant	-0.243***	0.502***	(0.017)	(0.024)
	No. Observations	. ,	. ,	13	3,340
Engogenenty rest (D-value) 0.00590 0.0348	F-stat Endogeneity Test (p-value)		254.1 00590		62.4 0348

 $\frac{38}{\text{Notes. Robust standard errors clustered at hospital level in parentheses. Results obtained using a sample of first-time and mulitple-time mothers. *** p<0.01, ** p<0.05 and * p<0.10.$

	Outcome Variable: Postnatal Depression			
	Linear IV model Bivariate probit model 2			
	(1)	(2)	(3)	
Unplanned CS	0.188***	0.116^{*}	0.406	
	(0.069)	(0.006)	(0.253)	
N.observations	$5,\!896$	5,896	5,896	

Table 11: Alternative model specifications

Notes. Standard errors in parentheses. In all the models, we use breech position as exclusion restriction. Linear IV models estimated using two-stages least squares (2SLS). Standard errors from the two-stage residual inclusion (2SRI) specification are obtained using bootstrapping. Column 2 reports the average marginal effects estimated from the bivariate probit model using the Stata command margins. Standard errors in the bivariate probit models are estimated using the deltamethod. All model specifications include information on socio-economic status, pregnancy, mother's health and baby's health as in column (3) of Table 6.*** p<0.01, ** p<0.05 and * p<0.10.

	Outcome Variable: Postnatal Depression			
	Linear IV model excluding baby's health measures		Linear IV model using alternative baby's health measures	
	without FE (1)	with FE (2)	without FE (3)	with FE (4)
Unplanned CS	0.186***	0.149**	0.186***	0.149**
Hospitalisation	(0.069)	(0.068)	(0.068) 0.045^{**} (0.021)	(0.068) 0.049^{**} (0.021)
No. Observations	5,896	5,864	5,896	5,864
F-test (first stage) Endogeneity Test (p-value)	$160.6 \\ 0.0137$	$163.7 \\ 0.0614$	$161.1 \\ 0.0124$	163.9 0.0568

Table 12: Robustness check: Baby's health measures

Notes. Robust standard errors clustered at hospital level in parentheses. All model specifications include information on socio-economic status, pregnancy and mother's health as in column (2) of Table 6. The alternative measure of baby's health is a dummy indicating whether the baby was hospitalised in the first 9 months after birth. *** p<0.01, ** p<0.05 and * p<0.10.

	Outcome Variable: Postnatal Depression			
	Linear IV model			
	without FE	with FE		
	(1)	(2)		
Effect of unplanned CS for:				
High educated mothers	0.232^{*}	0.207		
	(0.131)	(0.135)		
Low educated mothers	0.164	0.121		
	(0.102)	(0.100)		
No. Observations	5,896	5,864		
F-stat	25.82	27.29		
Endogeneity Test p-value	0.0343	0.131		

Table 13: Heterogeneity by mother's level of education

Notes. Robust standard errors clustered at hospital level in parentheses. All model specifications include information on socio-economic status, pregnancy and mother's and baby's as in column (3) of Table 6. High-educated mothers have a university qualification, while low-educated mothers have lower education (if any). *** p<0.01, ** p<0.05 and * p<0.10.